2019070713 8/12/2019 2:04 PM PGS 58 \$298.00 DF \$0.00 Electronically Recorded Jefferson County, CO George P Stern, Clerk and Recorder TD1000 N

DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS

FOR

CANDELAS PATIO HOMES

(A Residential Planned Community)

THIS DECLARATION CONTAINS MANDATORY ALTERNATIVE DISPUTE RESOLUTION PROVISIONS, IN LIEU OF LITIGATION, THAT CANNOT BE AMENDED OR DELETED WITHOUT DECLARANT CONSENT

AFTER RECORDING, RETURN TO:

Remington Homes at Candelas LLC 5740 Olde Wadsworth Boulevard Arvada, Colorado 80002

TABLE OF CONTENTS

PREA	MBLE	
ARTI	CLE ONE DEFINITIONS	2
ARTI	CLE TWO SCOPE OF THE DECLARATION	6
2.1	Property Subject to this Declaration	
2.2	Conveyances Subject to this Declaration	6
2.3	Owner's Rights Subject to Governing Documents	6
2.4	Initial Lots	6
2.5	Identification and Boundaries of Lots	6
ARTI	CLE THREE COMMON AREAS	6
3.1	Transfer of Title to the Common Areas	6
3.2	Owner's Rights in the Common Areas	6
3.3	Delegation of Use	7
ARTIC	CLE FOUR THE ASSOCIATION	8
4.1	Name	
4.2	Purposes and Powers	
4.3	Board/Managing Agent	9
4.4	Articles and Bylaws	9
4.5	Membership	9
4.6	Voting Rights	9
4.7	Declarant Control of the Association	9
4.8	Election by Owners	9
4.9	Budget	10
4.10	Association Agreements	
4.11	Indemnification	10
ARTIC	CLE FIVE ASSESSMENTS	10
5.1	Obligation	
5.2	Purpose of the Common Expense Assessments	11
5.3	Date of Commencement of Common Expense Assessments; Declarant's	
	Right of Offset	11
5.4	Levy of Assessments	11
5.5	Due Date	12
5.6	Remedies for Nonpayment of Assessments	12
5.7	Assessment Lien	13
5.8	Surplus Funds	
5.9	Initial Association Payment	13
5.10	Certificate of Assessment Status	

ARTI	CLE SIX USE RESTRICTIONS	1
6.1	Limitations and Restrictions	14
6.2	Residential Use	14
6.3	Animals and Pets	14
6.4	Nuisances	14
6.5	No Hazardous Activities	14
6.6	Compliance with Insurance Requirements	14
6.7	Owner Caused Damages	14
6.8	Lease of a Lot	15
ARTI	CLE SEVEN EASEMENTS	1.6
7.1	Generally	10 1 <i>6</i>
7.2	Utility Easements	10
7.3	Easements for Association and Owners	10 1 <i>6</i>
7.4	Emergency Easements	17
7.5	Recording Data Regarding Easements	17
7.6	Expansion Property Easement.	
7.7	Easements Deemed Appurtenant	1 / 1 7
7.8	Easements for Encroachments	17
ARTI	CLE EIGHT INSURANCE	10
8.1	Association Insurance	1 0
8.2	Separate Insurance	ه ۱ م
8.3	Owner Insurance	20
ARTI	CLE NINE RESTORATION UPON DAMAGE OR DESTRUCTION/	
	CONDEMNATION	20
9.1	Duty to Restore Common Areas	20
9.2	Use of Insurance Proceeds	20
9.3	Condemnation	20
ARTI	CLE TEN MAINTENANCE	20
10.1	Maintenance of Common Areas and Association Maintenance Areas	20
10.2	Maintenance of the Lots	21
10.3	Owner's Failure to Maintain and Repair	21
10.4	Maintenance of the Drainage Pattern/Irrigation System	21
10.5	Owner-Caused Damages	21
ARTI		
11.1	Reservation of Right to Expand	21
11.2	Reservation of Right to Expand	21
11.3	Supplemental Declarations.	22
11.4	Expansion of Definitions	22
11.5	Interests on Enlargement Termination of the Right of Expansion	22
	or the regit of Lapanoith	

ARTI	CLE TWELVE DECLARANT RIGHTS	23
12.1	Reservation	22
12.2	Rights Transferable	23
12.3	Limitations	23
12.4	Interference with Declarant Rights	24
12.5	Declarant's Easements	2/
12.6	Exercise of Rights	24
ARTIC	CLE THIRTEEN DESIGN REVIEW	24
13.1	Generally	24
13.2	Committee Approval of Improvements Required	25
13.3	Membership of the Committee	25
13.4	Address of the Committee	25
13.5	Submission of Plans/Design Review Fee	25
13.6	Delegation/Waiver	26
13.7	Criteria for Approval	26
13.8	Decision of the Committee	26
13.9	Appeal to the Board	26
13.10	Failure of Committee to Act on Plans	27
13.11	Prosecution of Work after Approval	27
13.12	Notice of Completion	27
13.13	Inspection of Work	27
13.14	Letter of Noncompliance	27
13.15	Failure of Committee to Act after Completion	28
13.16	Appeal to the Board of a Finding of Noncompliance	28
13.17	Correction of Noncompliance	28
13.18	Meetings of the Committee	28
13.19	No Implied Waiver or Estoppel	28
13.20	Estoppel Certificates	28
13.21	No Liability for Committee Action	29
13.22	Exemptions for Declarant	29
ARTIC	CLE FOURTEEN MORTGAGEE PROVISIONS	29
14.1	Notice to Eligible Mortgagees	29
14.2	Action by Mortgagee	29
ARTIC	CLE FIFTEEN OWNER'S ACKNOWLEDGMENTS AND WAIVERS	29
15.1	Mold	
15.2	Noise Disturbances	
15.3	No View Easement	30
15.4	Security	31
15.5	Other Properties	31
15.6	Soils Condition	31
15.7	Inspection by Others; Waiver of Post Inspection Liability	32

ARTI	CLE SIXTEEN DISPUTE RESOLUTION	33
16.1	Definitions Applicable to this Article Sixteen	
16.2	Intent of Parties; Applicability of Article; and Applicability of Statutes of	
	Limitations	34
16.3	Exclusions from "Claim	34
16.4	Notice; Right to Inspect and Correct; Mediation	35
16.5	Commencing a Formal Claim.	3 <i>6</i>
16.6	Notice of Certain Claims	40
16.7	Amendment	40 40
16.8	Waiver of Jury Trial	40 40
16.9	Conflict with Laws	41
ARTI	CLE SEVENTEEN DURATION, AMENDMENT, AND TERMINATION	
	OF THE DECLARATION	41
17.1	Duration	
17.2	Amendments by Owners	41
17.3	Amendments by Declarant	41
17.4	Termination	41
17.5	Consent Required	42
ARTI	CLE EIGHTEEN GENERAL PROVISIONS	42
18.1	Right of Action	42
18.2	Successors and Assigns	
18.3	Severability	43
18.4	No Waiver	43
18.5	Registration by Owner of Mailing Address; Notices	43
18.6	Conflicting Provisions	43
18.7	Captions	43
18.8	Numbers and Genders	44
MORT	GAGEE CONSENT	46
EXHII	BIT A	47
	BIT B	
EXHII		40

DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR CANDELAS PATIO HOMES

THIS DECLARATION OF COVENANTS, CONDITIONS AND RESTRICTIONS FOR CANDELAS PATIO HOMES ("<u>Declaration</u>") is made as of August ______, 2019, by Remington Homes at Candelas LLC, a Colorado limited liability company ("<u>Declarant</u>").

PREAMBLE

WHEREAS, Declarant is the owner of certain real property located in the County of Jefferson, State of Colorado, more particularly described on the attached **Exhibit A** ("**Property**");

WHEREAS, Declarant also is, or will be, the owner of certain real property located in the County of Jefferson, State of Colorado, more particularly described on the attached **Exhibit B** ("**Expansion Property**");

WHEREAS, Declarant intends to create a planned community on the Property, and Declarant wishes for the Property to become a part of such planned community;

WHEREAS, Declarant reserves the right to annex all or portions of Expansion Property into said planned community from time to time; and

WHEREAS, the Property will be conveyed subject to the protective covenants, conditions and restrictions as hereinafter set forth.

NOW THEREFORE, Declarant hereby submits the Property, together with all rights and appurtenances thereto and improvements thereon, to the terms and provisions hereof and of the Colorado Common Interest Ownership Act ("Act") in order to form a planned community called "Candelas Patio Homes." Declarant hereby declares that all of said Property shall be held and conveyed subject to the following covenants, conditions and restrictions, all of which are declared and agreed to be for the protection of the value of the Property and for the benefit of any Person having any right, title or interest in the Property. Said covenants, conditions and restrictions shall be deemed to run with the land and shall be a burden and a benefit to any Person acquiring such interest, their grantees, heirs, legal representatives, successors and assigns, and acceptance of such interest by any such Person shall constitute such Person's agreement to be bound by the same.

The covenants, conditions and restrictions set forth in this Declaration are in addition to those set forth in the Protective Covenants and Easements of Candelas recorded with the Clerk and Recorder on March 26, 2012 at Reception No. 2012032029, as amended and supplemented from time to time ("Master Declaration").

THE TERMS AND PROVISIONS OF THIS DECLARATION REQUIRING ALTERNATIVE DISPUTE RESOLUTION FOR CONSTRUCTION DEFECT CLAIMS INURE TO THE BENEFIT OF DECLARANT AND THE DEVELOPMENT PARTIES (AS DEFINED IN SECTION 16.1

BELOW), ARE ENFORCEABLE BY DECLARANT AND SHALL NOT BE AMENDED FOR A PERIOD OF 20 YEARS FROM THE DATE THIS DECLARATION IS RECORDED WITH THE CLERK AND RECORDER WITHOUT THE WRITTEN CONSENT OF DECLARANT, WITHOUT REGARD TO WHETHER DECLARANT OWNS ANY PORTION OF THE PROJECT AT THE TIME OF SUCH AMENDMENT. BY TAKING TITLE TO A LOT, ALL OWNERS AND THE ASSOCIATION ACKNOWLEDGE AND AGREE THAT DECLARATION PROVISIONS REQUIRING ALTERNATIVE DISPUTE RESOLUTION OF CONSTRUCTION DEFECT CLAIMS ARE A SIGNIFICANT INDUCEMENT TO THE DECLARANT'S WILLINGNESS TO DEVELOP AND SELL THE LOTS AND THAT IN THE ABSENCE OF THE ALTERNATIVE DISPUTE RESOLUTION PROVISIONS CONTAINED IN THIS DECLARATION, DECLARANT WOULD HAVE BEEN UNABLE AND UNWILLING TO DEVELOP AND SELL THE LOTS FOR THE PRICES PAID BY THE ORIGINAL PURCHASERS.

ARTICLE ONE DEFINITIONS

As used in this Declaration, unless the context otherwise requires, the terms hereinafter set forth shall have the following meanings:

- 1.1 "Act" means the Colorado Common Interest Ownership Act, C.R.S. §38-33.3-101, et seq.
- 1.2 "<u>Agency</u>" means any agency or corporation such as the U.S. Department of Housing and Urban Development ("<u>HUD</u>"), U.S. Veterans' Administration ("<u>VA</u>"), Federal National Mortgage Association ("<u>FNMA</u>") or the Federal Home Loan Mortgage Corporation ("<u>FHLMC</u>") that purchases or insures residential mortgages.
- 1.3 "<u>Allocated Interests</u>" means the Common Expense Liability and the votes in the Association that are allocated to each of the Lots in the Project. The formulas used to establish the Allocated Interests are as follows:
- (a) Common Expense Liability. Subject to Section 5.4 below, all Common Expenses shall be levied against Lots on the basis of a fraction, the numerator of which is one and the denominator of which is the total number of Lots then within the Project.
 - (b) Votes. Owners shall be entitled to one vote for each Lot owned within the Project.
- 1.4 "Articles" means the Articles of Incorporation of the Association.
- 1.5 "<u>Assessments</u>" means the (a) Common Expense Assessments, (b) Special Assessments, (c) Individual Assessments, (d) Fines and (e) Costs of Enforcement levied pursuant to this Declaration.
- 1.6 "Assessment Lien" means the statutory lien on a Lot for any Assessment levied against that Lot, as described in Section 5.7 below.
- 1.7 "<u>Association Maintenance Areas</u>" means those portions of the Lots to be maintained by the Association as further described in the Rules.

- 1.8 "<u>Association</u>" means the Candelas Patio Homes Homeowners Association, Inc., a Colorado nonprofit corporation, its successors and assigns.
- 1.9 "**Board**" means the Board of Directors of the Association duly elected pursuant to the Bylaws of the Association or appointed by Declarant as provided in Section 4.7 below.
- 1.10 "<u>Budget</u>" means the annual budget of the Association prepared and adopted in accordance with Section 4.9 below.
- 1.11 "<u>Bylaws</u>" means the Bylaws which are adopted by the Board for the regulation and management of the Association.
- 1.12 "<u>City</u>" means the City of Arvada, State of Colorado.
- 1.13 "Clerk and Recorder" means the office of the Clerk and Recorder in the County.
- 1.14 "<u>Common Areas</u>" means any real property (including all improvements located thereon), which is owned by the Association for the common use and enjoyment of the Owners unless otherwise provided herein (e.g., Limited Common Areas defined below). Initially there are no Common Areas within the Project.

Notwithstanding the foregoing, certain of the Common Areas may be "<u>Limited Common Areas</u>" which are either limited to or reserved in this Declaration, in a recorded amendment to this Declaration executed by Declarant pursuant to Article Twelve below, or by authorized action of the Association, for the exclusive use of an Owner, or are limited to and reserved for the common use of more than one but fewer than all Owners.

- 1.15 "Common Expense Assessments" means the funds required to be paid by each Owner in payment of such Owner's Common Expense Liability as more fully defined in Section 5.2 below.
- 1.16 "<u>Common Expense Liability</u>" means the liability for the Common Expense Assessments allocated to each Lot and determined in accordance with that Lot's Allocated Interests as set forth in Section 1.3(a) above.
- 1.17 "<u>Common Expenses</u>" means expenditures made by or liabilities incurred by or on behalf of the Association, together with allocations to reserve funds.
- 1.18 "Costs of Enforcement" means all fees, late charges, interest and expenses, including receivers' fees, reasonable attorneys' fees and costs incurred by the Association in connection with the collection of Assessments, and the enforcement of the terms, conditions and obligations of the Governing Documents. Costs of Enforcement are collectible and enforceable as Assessments.
- 1.19 "County" means Jefferson County, Colorado.
- 1.20 "<u>Declarant</u>" means Remington Homes at Candelas LLC, a Colorado limited liability company, and its successors and assigns. A Person shall be deemed a "successor and assign" of Declarant only if specifically designated in a duly recorded instrument as a successor or assign of

Declarant under this Declaration, and shall be deemed a successor and assign of Declarant only as to the particular rights or interests of the Declarant under this Declaration which are specifically designated in the recorded instrument.

- 1.21 "<u>Declarant Rights</u>" means the development rights, special declarant rights and other rights granted to or reserved by Declarant as set forth in this Declaration and the Act.
- 1.22 "<u>Declaration</u>" means this Declaration, the Plat and any supplements and amendments thereto recorded with the Clerk and Recorder.
- 1.23 "<u>Eligible Mortgagee</u>" means a holder, insurer or guarantor of a First Mortgage who has delivered a written request to the Association containing its name, address, the legal description and the address of the Lot encumbered by its First Mortgage, requesting that the Association notify them on any proposed action requiring the consent of the specified percentage of Eligible Mortgagees.
- 1.24 "Fines" means those fines described in Section 5.4(d) below.
- 1.25 "<u>First Mortgage</u>" means any Mortgage the priority of which is not subject to any monetary lien or encumbrance except liens for taxes or other liens that are given priority by statute.
- 1.26 "Governing Documents" means this Declaration, the Plat, the Articles, the Bylaws and the Rules, if any (and any other documents promulgated by the Board), as they may be amended or supplemented from time to time.
- 1.27 "Guest" means (a) any Person who resides with an Owner within the Project, (b) a guest or invitee of an Owner, (c) an occupant or tenant of a Home within the Project, and any members of his or her household, invitee or cohabitant of any such Person, or (d) a contract purchaser.
- 1.28 "<u>Home</u>" means the residence constructed on each Lot within the Project (including any garage serving such residence) and any replacements thereof.

1.29 "Improvements" means:

- (a) all exterior improvements, structures and any appurtenances thereto or components thereof of every type or kind; and
- (b) the grading, excavation, filling or similar disturbance to the surface of the land including, without limitation, change of grade, change of ground level, and change of drainage pattern; and
- (c) all landscaping and hardscaping features, including, but not limited to, buildings, outbuildings, patios, patio covers, awnings, painting or other finish materials on any visible structure, additions, walkways, sprinkler systems, garages, private drives, driveways, fences, screening walls, retaining walls, stairs, decks, landscaping, hedges, windbreaks, plantings, trees, shrubs, flowers, vegetables, sod, gravel, bark, exterior light fixtures, poles, signs, cooling, heating and water softening equipment; and

- (d) any change, alteration, modification, expansion, or addition to any previously approved Improvement, including any change of exterior appearance, finish material, color or texture.
- 1.30 "Individual Assessments" means those Assessments defined in Section 5.4(c) below.
- 1.31 "<u>Lot</u>" means each lot shown on the Plat which is subject to this Declaration, together with all appurtenances and improvements now or hereafter located thereon. "Lot" shall include any Home constructed thereon as the term Home is defined above.
- 1.32 "<u>Lots That May Be Created</u>" means 99 Lots, including those Lots which may be included if all of the real property described on <u>Exhibit B</u> is annexed into the Project and made subject to this Declaration.
- 1.33 "<u>Managing Agent</u>" means any one or more persons employed by the Association who is engaged to perform any of the duties, powers or functions of the Association.
- 1.34 "Member" means each Owner.
- 1.35 "<u>Mortgage</u>" means any mortgage, deed of trust or other document conveying any Lot or interest therein to a Mortgagee, but only as security for payment of a debt or obligation and not intended to initially convey fee simple title thereof.
- 1.36 "Mortgagee" means any Person named as a mortgagee or beneficiary in any Mortgage, or any successor to the interest of any such Person under such Mortgage.
- 1.37 "<u>Notice and Hearing</u>" means a written notice and an opportunity for a hearing before the Board in the manner provided for in the Bylaws.
- 1.38 "Owner" means the record owner of the fee simple title to any Lot which is subject to this Declaration.
- 1.39 "Owner's Agent" means any agent, contractor, employee, family member, licensee or Guest of an Owner.
- 1.40 "Period of Declarant Control" means that period of time as defined in Section 4.7 below.
- 1.41 "Person" means a natural person, a corporation, a partnership, an association, a trustee, a limited liability company, a joint venture, or any other entity recognized as being capable of owning real property under Colorado law.
- 1.42 "<u>Plat</u>" means Candelas Filing No. 4, Amendment No. 1, Final Plat, recorded with the Clerk and Recorder on October 12, 2017, at Reception No. 2017105258, and any supplements or amendments thereto.

- 1.43 "<u>Project</u>" means the common interest community created by this Declaration and as shown on the Plat consisting of the Property, and such additional property that may be annexed into the common interest community pursuant to Article Eleven below.
- 1.44 "Rules" means the rules and regulations adopted by the Board for the regulation and management of the Project, if any, as amended from time to time.
- 1.45 "Special Assessments" means those Assessments defined in Section 5.4(b) below.

ARTICLE TWO SCOPE OF THE DECLARATION

- 2.1 <u>Property Subject to this Declaration</u>. Declarant as the owner of title to the Property, does hereby subject the Property to the provisions of this Declaration thereby creating a planned community.
- 2.2 <u>Conveyances Subject to this Declaration</u>. All covenants, conditions and restrictions which are granted or created by this Declaration shall be deemed to be covenants appurtenant to and running with the land, and shall at all times inure to the benefit of and be binding on any Person having any interest in the Project, their respective heirs, successors, personal representatives or assigns, except as otherwise provided herein.

Any instrument recorded subsequent to this Declaration and purporting to establish and affect any interest in the Project shall be subject to the provisions of this Declaration despite any failure to make reference thereto.

- 2.3 Owner's Rights Subject to Governing Documents. Each Owner shall own their Lot subject to the provisions of the Governing Documents.
- 2.4 <u>Initial Lots</u>. The initial number of Lots within the Project are set forth on the attached <u>Exhibit A</u>. Declarant reserves the right, but not the obligation, to add and/or create additional Lots by the expansion of the Project in accordance with Article Eleven below.
- 2.5 <u>Identification and Boundaries of Lots</u>. The identification number and boundaries of each Lot is as shown on the Plat.

ARTICLE THREE COMMON AREAS

- 3.1 <u>Transfer of Title to the Common Areas</u>. Upon the annexation of Common Areas into the Project (if any), Declarant will convey title thereto to the Association. Any property conveyed to the Association shall be conveyed free and clear of all monetary liens and monetary encumbrances (other than the lien of real estate taxes not then due and payable), but otherwise subject to all covenants, conditions, easements and restrictions of record.
- 3.2 Owner's Rights in the Common Areas. Subject to use restrictions contained herein, or in any other recorded instrument affecting the Project, or promulgated by the Association, every Owner and

their Guests shall have the right and easement of use and enjoyment in and to the Common Areas which shall be appurtenant to and shall pass with title to the Lot to such Owner; *provided, however*, that the right to use and enjoyment of any Limited Common Areas are limited to the Owners of the Lots to which such Limited Common Areas are allocated pursuant to this Declaration. Each Owner's right to own and use their Lot and the Common Areas shall also be subject to the Declarant Rights reserved herein and the following rights of the Board:

- (a) To borrow money to improve the Common Areas and to mortgage said Common Areas as security for any such loan; *provided*, *however*, that the Association may not subject any portion of the Common Areas to a security interest unless approved by (i) Owners to which at least 67% of the votes in the Association are allocated, including 67% of the votes allocated to Lots not owned by Declarant, and (ii) all Owners to which any Limited Common Area is allocated agree in order to subject that Limited Common Area to a security interest, as more fully set forth in §312 of the Act.
- (b) To convey or dedicate all or any part of the said Common Areas for such purposes and subject to such conditions as may be agreed to by (i) the Owners to which at least 67% of the votes in the Association are allocated, including 67% of the votes allocated to Lots not owned by Declarant, and (ii) all Owners of any Limited Common Area conveyed or dedicated as more fully set forth in §312 of the Act. The granting of permits, licenses and easements shall not be deemed a conveyance or dedication within the meaning of this Section 3.2(b).
- (c) To promulgate and adopt Rules with which each Owner and their Guests shall strictly comply.
- (d) To suspend the voting rights of an Owner for any period during which any Assessment remains unpaid or for any violation of a Governing Document.
- (e) To take such steps as are reasonably necessary to protect the Common Areas against foreclosure.
- (f) To enter into, make, grant, perform or enforce any contracts, leases, agreements, licenses, easements and rights-of-way for the use of Common Areas by Owners, Guests and any other Persons for any purpose the Board may deem to be useful, beneficial or otherwise appropriate.
- (g) To close or limit the use of the said Common Areas temporarily while maintaining, repairing or replacing Common Areas.
- (h) To make such use of the Common Areas as may be necessary or appropriate for the performance of the duties and functions which it is obligated or permitted to perform under this Declaration.
 - (i) The rights granted to the Board in Section 4.2 below.
- 3.3 <u>Delegation of Use</u>. Any Owner may delegate their right of enjoyment of the Common Areas to their Guests subject to limitations created by the Board.

ARTICLE FOUR THE ASSOCIATION

- 4.1 <u>Name</u>. The name of the Association is the Candelas Patio Homes Homeowners Association, Inc., and it is a Colorado nonprofit corporation.
- 4.2 <u>Purposes and Powers</u>. The Association, through its Board, shall manage, operate, care for, insure, maintain, repair and reconstruct all of the Common Areas and keep the same in an attractive and desirable condition for the use and enjoyment of all of the Owners and their Guests. Any purchaser of a Lot shall be deemed to have assented to, ratified and approved such designations and management rights. The Board shall have all of the powers, authority and duties permitted pursuant to the Act necessary and proper to manage the business and affairs of the Association, including, but not limited to, the following:
- (a) Attorney-in-Fact. The Board is hereby irrevocably appointed attorney-in-fact for each Owner in order (i) to manage, control and deal with the interest of each Owner in the Common Areas, so as to permit the Association to fulfill all of its duties and obligations hereunder and to exercise all of its rights hereunder, and (ii) to deal with the Project upon its destruction, condemnation or obsolescence as hereinafter provided.

Acceptance of any interest in any Lot shall constitute an appointment of the Board as attorney-in-fact as provided above and hereinafter. The Board shall be granted all of the powers necessary to govern, manage, maintain, repair, administer and regulate the Project and to perform all of the duties required of it.

(b) Contracts, Easements, and Other Agreements. Subject to Section 4.10 below, the Board shall have the right to enter into, grant, perform, enforce, cancel and vacate any contracts, easements, licenses, leases, agreements, and/or rights-of-way, for the use of Common Areas by Owners, Guests or any other Persons, including, without limitation, the general public.

Any of such contracts, easements, licenses, leases, agreements and/or rights-of-way, shall be upon such terms and conditions as may be agreed to from time to time by the Board, without the necessity of the consent thereto, or joinder therein, of the Owners, Mortgagees or any other Person.

- (c) Other Association Functions. The Board may undertake any activity, function or service for the benefit of or to further the interests of all, some or any Owners on a self-supporting, Common Expense Assessment or Special Assessment basis.
- (d) Implied Rights. The Board shall have and may exercise any right or privilege given to it expressly by this Declaration, or reasonably to be implied from the provisions of this Declaration, or given or implied by law, including, without limitation, the Act and the Colorado Revised Nonprofit Act ("Nonprofit Act"), or which may be necessary or desirable to fulfill its duties, obligations, rights or privileges.

- 4.3 <u>Board/Managing Agent</u>. The affairs of the Association shall be managed by the Board. By resolution, the Board may delegate authority to a Managing Agent for the Association as more fully provided for in the Bylaws, provided no such delegation shall relieve the Board of final responsibility.
- 4.4 <u>Articles and Bylaws</u>. The purposes and powers of the Association and the rights and obligations with respect to Owners set forth in this Declaration may and shall be amplified by provisions of the Articles and Bylaws.
- 4.5 <u>Membership</u>. Members of the Association shall be every record Owner of a Lot subject to this Declaration. Membership shall be appurtenant to and may not be separated from ownership of any Lot. Where more than one Person holds an interest in any Lot, all such Persons shall be Members but there shall be only one vote per Lot.

The membership of the Association at all times shall consist exclusively of all Owners or, following termination of the Project, of all former Owners entitled to distributions of the proceeds under §218 of the Act, or their heirs, personal representatives, successors or assigns.

- 4.6 <u>Voting Rights</u>. The Association shall have one class of voting membership. Owners shall be entitled to one vote in the Association for each Lot owned within the Project.
- 4.7 <u>Declarant Control of the Association</u>. Subject to the provisions of Section 4.8 below, there is a "<u>Period of Declarant Control</u>" during which period Declarant may appoint and remove any officer of the Association or any member of the Board. The Period of Declarant Control starts upon the recording of this Declaration and terminates no later than the earlier of:
- (a) 60 days after conveyance by Declarant of 75% of the Lots That May Be Created to Owners other than a Declarant;
- (b) 2 years after the last conveyance of a Lot by a Declarant in the ordinary course of business; or
 - (c) 2 years after any right to add new Lots to the Project was last exercised by Declarant.

Declarant may voluntarily surrender the right to appoint and remove officers of the Association and members of the Board before termination of the Period of Declarant Control in a recorded instrument executed by Declarant. In that event, Declarant may require, for the duration of the Period of Declarant Control, that specified actions of the Board, as described in such recorded instrument, be approved by Declarant before they become effective.

4.8 <u>Election by Owners</u>. Not later than 60 days after the conveyance of 25% of the Lots That May Be Created to Owners other than a Declarant, at least one member and not less than 25% of the members of the Board must be elected by Owners other than Declarant.

Not later than 60 days after conveyance of 50% of the Lots That May Be Created to Owners other than a Declarant, not less than 33 1/3% of the members of the Board must be elected by Owners other than Declarant.

Not later than the termination of the Period of Declarant Control, Owners shall elect a Board consisting of at least 3 members, at least a majority of who must be Owners other than Declarant. The Board shall elect the officers of the Association.

4.9 Budget.

- (a) Annual Budget. Within 90 days after the adoption of any Budget by the Board, the Board shall mail, by ordinary first-class mail, or otherwise deliver, a summary of the Budget to each Owner and shall set a date for a meeting of the Owners to consider ratification of the Budget within a reasonable time after delivery of the summary. Unless at that meeting Owners to which at least a majority of all of the votes in the Association are allocated reject the Budget, the Budget shall be deemed ratified whether or not a quorum is present. In the event the Budget is rejected, the Budget last ratified by the Owners must be continued until such time as the Owners ratify a subsequent Budget adopted by the Board.
- (b) Amended Budget. If the Board deems it necessary or advisable to amend a Budget that has been ratified by the Owners pursuant to Section 4.9(a) above, the Board may adopt a proposed amendment to the Budget, deliver a summary of the proposed amendment to all Owners and set a date for a meeting of the Owners to consider ratification of the proposed amendment. The date of such meeting shall be a reasonable time after the delivery of the summary of the proposed amendment. Unless at that meeting Owners to which at least a majority of the votes in the Association are allocated reject the amended Budget, the amended Budget shall be deemed ratified whether or not a quorum is present.
- 4.10 <u>Association Agreements</u>. Any agreement for professional management of the Project, or any contract providing for services of Declarant, may be terminated without penalty to the Association at any time after termination of the Period of Declarant Control, upon not less than 90 days-notice to the other party.
- 4.11 <u>Indemnification</u>. Each officer of the Association and each director of the Board shall be indemnified by the Association against all expenses and liabilities including attorneys' fees, reasonably incurred by or imposed upon him or her in any proceeding to which he or she may be a party, or in which he or she may become involved, by reason of his or her being or having been an officer of the Association or director of the Board, or any settlements thereof, whether or not he or she is an officer of the Association or director of the Board at the time such expenses are incurred, to the full extent permitted by Colorado law and as further set forth in the Bylaws. In the event of any conflict between the terms of this Section 4.11 and the Bylaws, the Bylaws shall control.

ARTICLE FIVE ASSESSMENTS

5.1 <u>Obligation</u>. Each Owner shall be personally obligated to pay to the Association (a) Common Expense Assessments, (b) Special Assessments, (c) Individual Assessments, (d) Fines and (e) Costs of Enforcement, which shall be a continuing lien upon the Lot against which each such Assessment is levied. The phrase "personally obligated" as used herein and in other provisions of this

Declaration, is not intended, and shall not convey personal liability on the members and/or shareholders of a corporation or limited liability company, including, without limitation, with respect to Declarant.

The obligation for such payments by each Owner to the Association is an independent personal covenant with all amounts due, from time to time, payable in full when due without notice or demand and without setoff or deduction. All Owners of each Lot shall be jointly and personally liable to the Association for the payment of all Assessments attributable to their Lot.

Each Assessment against a Lot is the personal obligation of the Person(s) who owned the Lot at the time the Assessment became due and shall also pass to successors in title. By acceptance of a deed for a Lot, each Lot purchaser thereby consents to assume the foregoing joint obligation for all Assessments due against the Lot pursuant to this Section 5.1.

The omission or failure of the Board to levy Assessments for any period shall not be deemed a waiver, modification or a release of the Owners from their obligation to pay Assessments.

No Owner may waive or otherwise escape liability for the payment of Assessments by the non-use of the Common Areas or the abandonment of such Owner's Lot.

- by the Association shall be used for the purpose of promoting the welfare and interests of the Owners, including, but not limited to (a) providing for the administration and management of the Project, (b) providing for the upkeep, operation, improvement, repair, maintenance and reconstruction of the Common Areas (and other areas of the Project required by this Declaration, including the Association Maintenance Areas), including, but not limited to, the maintenance required in Article Ten below, (c) providing the insurance required by Article Eight below, (d) creating a reasonable and adequate contingency or other reserve or surplus funds for insurance deductibles and general, routine maintenance, repairs and replacement of the Common Areas on a periodic basis, as needed, and (e) satisfying any other purpose reasonable, necessary or incidental to such purposes.
- Date of Commencement of Common Expense Assessments; Declarant's Right of Offset. The Common Expense Assessment shall commence as to the Lots no later than 60 days after the first Lot is conveyed to an Owner other than a Declarant. Until the commencement of the collection of the Common Expense Assessment, Declarant shall pay all of the expenses incurred and paid for by the Association. Declarant may at any time advance operating funds to the Association. Declarant shall be entitled to offset such amounts so paid or advanced as a credit against future Common Expense Assessments payable by Declarant.

5.4 <u>Levy of Assessments</u>.

(a) Common Expense Assessments. Common Expense Assessments shall be levied based upon the Budget's cash requirements. The Common Expense Liability shall be allocated among all Lots in accordance with each Lot's Common Expense Liability as set forth in Section 1.3(a) above and shall commence in accordance with Section 5.3 above.

To the extent that any Common Expenses, or a portion thereof benefit fewer than all Owners, such expenses may be assessed exclusively against the Lots benefited as provided in §315(3)(b) of the Act, as determined by the Board. For example, in the event that a Lot subjected to this Declaration is not (and has never been) occupied, the Board may determine to charge the Owner of such Lot an Assessment less than its Common Expense Liability described in Section 1.3(a).

- (b) Special Assessments. In addition to the other Assessments authorized herein, the Board, subject to the requirements set forth below, may levy a Special Assessment for the purpose of defraying, in whole or in part, any unexpected expense, including, but not be limited to, the cost of any construction, reconstruction, improvement, repair or replacement of the Common Areas, or for the funding of any operating deficit incurred by the Association provided that, except as otherwise set forth herein, any such Assessment shall have the approval of Owners to whom at least a majority of the votes in the Association are allocated. Any such Special Assessment shall be levied against each Lot in accordance with that Lot's Common Expense Liability determined in accordance with Section 1.3(a) above, subject to §315(3)(b) of the Act and Section 5.3 above.
- (c) Individual Assessments. The Board shall have the right to individually levy upon any Owner(s) amounts as provided for by this Declaration. Individual Assessments may be levied at any time as required and are exempt from any voting requirements by the Owners required by other Assessments called for under this Declaration.
- (d) Fines. The Board shall have the right to levy a Fine against an Owner(s) for violations of the Governing Documents. No such Fine shall be levied until the Owner(s) to be charged has been given a Notice and Hearing. Fines may be levied in a reasonable amount as determined from time to time by the Board in its discretion and uniformly applied. Fines may be levied at any time as required and are exempt from any voting requirements of the Owners required by other Assessments called for under the Declaration.
- 5.5 <u>Due Date</u>. Special Assessments, Individual Assessments, Costs of Enforcement and Fines shall be due and payable as established by the Board. Common Expense Assessments shall be due and payable in monthly installments, in advance, or in such frequency as the Board determines in its discretion from time to time. Written notice of all Assessments shall be sent to each Owner subject thereto specifying the type of Assessment, the amount and the date such Assessment is due.
- 5.6 Remedies for Nonpayment of Assessments. If any Assessment is not fully paid within 5 days after the same becomes due and payable, then interest shall accrue at the default rate set by the Board on any amount of the Assessment in default accruing from the due date until the date of payment, and the Board may also assess a late fee in an amount as determined in the Board's discretion. In addition, the Board may in its sole discretion:
- (a) Accelerate and declare immediately due and payable all unpaid installments of any Assessment payable for the balance of the fiscal year during which such default occurred;
- (b) Bring an action at law against any Owner personally obligated to pay the Assessment and obtain a judgment for the amounts due; and/or

(c) Proceed to foreclose its Assessment Lien against the Lot pursuant to the power of sale granted to the Association by this Declaration in the manner and form provided by Colorado law for foreclosure of real estate mortgages, subject to the last paragraph of Section 5.7 below.

An action at law or in equity by the Association against an Owner to recover a judgment for unpaid Assessments may be commenced and pursued by the Association without foreclosing or in any way waiving its Assessment Lien. The Association may require reimbursement for Costs of Enforcement without the necessity of commencing a legal proceeding.

5.7 <u>Assessment Lien</u>. The Association is hereby granted a lien ("<u>Assessment Lien</u>") against each Lot for any Assessment levied by the Board. If an Assessment is payable in installments, the full amount of the Assessment is a lien from the time the first installment thereof becomes due. The Assessment Lien shall be superior to all other liens and encumbrances on a Lot except as set forth in §316 of the Act. No sale, deed in lieu of foreclosure, nor cancellation or forfeiture shall relieve an Owner from continuing liability for any Assessment thereafter becoming due, nor from the lien thereof.

Recording of the Declaration constitutes record notice and perfection of the lien. No further recordation of any claim of lien for Assessments under this Article Five is required. However, the Board may prepare, and record with the Clerk and Recorder, a written notice setting forth the amount of the unpaid indebtedness, the name of the Owner of the Lot, and a description of the Lot. If a lien is filed, the cost thereof shall be considered a Cost of Enforcement.

The Assessment Lien shall be superior to any homestead exemption now or hereafter provided by the laws of the State of Colorado or any exemption now or hereafter provided by the laws of the United States. The acceptance of a deed to a Lot subject to this Declaration shall constitute a waiver of the homestead and any other exemption as against said Assessment Lien.

Notwithstanding the foregoing provisions of this Section 5.7, in order to comply with §316(11)(a) of the Act, the Association may not foreclose its lien against a Lot unless (a) the amount of Assessments and other amounts due to the Association is equal to or greater than 6 months' worth of Common Expense Assessments against the Lot, and (b) there is a recorded vote of the Board authorizing the foreclosure of the specific Lot being foreclosed. The Board authority to initiate a foreclosure cannot be delegated to the Managing Agent, an attorney or any other Person, nor can the Association have a policy of automatically requiring foreclosure when an Owner becomes 6-months delinquent.

- 5.8 <u>Surplus Funds</u>. Any surplus funds of the Association remaining at the close of the Association's fiscal year after payment of the Association's expenses and funding reserves may be retained by the Association as unallocated reserves and need not be credited to the Owners to reduce their future Assessment liability.
- 5.9 <u>Initial Association Payment</u>. To provide the Association with sufficient funds to cover the cost of initial expenses and any future expenses authorized by the Board for which there are insufficient budgeted funds, at the closing of the initial sale of a Lot to an Owner (other than to a Declarant), a non-refundable payment shall be made by such Owner to the Association in an amount

equal to 2 months of Common Expense Assessments. Amounts paid pursuant to this Section 5.9 do not constitute advance payments of Common Expense Assessments and do not relieve an Owner from making regular payments of Common Expense Assessments as they become due. In the event that Declarant makes such payment on behalf of any Owner, such amount shall be reimbursable to Declarant by such Owner upon the conveyance of title to the Lot to such Owner by Declarant.

5.10 <u>Certificate of Assessment Status</u>. The Association shall furnish to an Owner or such Owner's Mortgagee(s) upon written request delivered personally or by certified mail, first class postage prepaid, return receipt requested, to the Managing Agent, if any, and if there is no Managing Agent then to the Board, a written statement setting forth the amount of unpaid Assessments currently levied against such Owner's Lot. The statement shall be furnished within 14 calendar days after receipt of the request and is binding upon the Association, the Board and every Owner. If no statement is furnished to the Owner or Mortgagee(s), delivered personally or by certified mail, first class postage prepaid, return receipt requested to the inquiring party, then the Association shall have no right to assert an Assessment Lien upon the Lot for unpaid Assessments which were due as of the date of the request.

ARTICLE SIX USE RESTRICTIONS

- 6.1 <u>Limitations and Restrictions</u>. All Lots and Common Areas shall be used and enjoyed subject to the following limitations and restrictions, and any other limitations and restrictions affecting the Project. Notwithstanding the foregoing, so long as Declarant owns a Lot within the Project, Declarant shall be exempt from the provisions of this Article Six to the extent that it impedes, in Declarant's sole discretion, its development, construction, marketing, sales or leasing activities.
- Residential Use. Each Lot shall be used and occupied primarily as a residence. No industry, business, trade, occupation, or profession of any kind, commercial, religious, educational, or otherwise, designed for profit, altruism, exploration, or otherwise, shall be conducted, maintained, or permitted in on any Lot except as expressly provided herein. The foregoing shall not, however, be construed in such a manner as to prohibit an Owner from (a) maintaining their personal professional library in a Home, (b) keeping their personal business or professional records or accounts in a Home, (c) handling their personal business or professional telephone calls or correspondence from a Home, (d) maintaining a computer or other office equipment within a Home, or (e) utilizing administrative help or meeting with business or professional associates, clients or customers in a Home. Any accessory business use of a Home, permitted by this Section 6.2, must be in compliance with all applicable statutes, ordinances and governmental regulations and must be conducted in accordance with the Governing Documents.
- 6.3 Animals and Pets. No household pet or animal shall be allowed in or about the Project at any time without close supervision by an Owner. Owners shall be responsible for strict compliance with all laws related to pet ownership, including any regulation wholly excluding or limiting the number or type or breed of pets allowed, and each such Owner shall ensure that their pet does not interfere with other Owners' quiet use and enjoyment of the Project. Owners will be held responsible for any litter, waste, mess, damage or offensive noises created by their pets. An Owner's right to keep household pets shall be coupled with the responsibility to pay for any damage caused by such pet(s).

- Nuisances. No noxious or offensive activity shall be carried on upon any part of the Project, nor shall anything be done or placed on or in any part of the Project that is or may become a nuisance or cause embarrassment, disturbance or annoyance to others. No activity shall be conducted on any part of the Project and no improvements shall be made or constructed on any part of the Project that are or might be unsafe or hazardous to any Person or property. No sound shall be emitted on any part of the Project that is unreasonably loud or annoying. No Owner shall operate any machines, appliances, electronic devices, sound systems, accessories or equipment in such a manner as to cause an unreasonable disturbance to others, or cause any damage to or overloading of any mechanical, electrical, plumbing, or any other system serving the Project. No odor that is unreasonably noxious or offensive to others shall be emitted on any part of the Project. Habitually barking, howling or yelping dogs shall be deemed a nuisance.
- 6.5 <u>No Hazardous Activities</u>. No activity shall be conducted on any portion of the Project which is or might be unsafe or hazardous to any person or property. Without limiting the generality of the foregoing, no firearms shall be discharged upon any portion of the Project and no open fires shall be lighted or permitted on any portion of the Project except in a contained barbecue unit while attended and in use for cooking purposes.
- 6.6 <u>Compliance with Insurance Requirements</u>. Except as may be approved in writing by the Board, nothing shall be done or kept on the Project which may result in an increase in the rates of insurance or would result in the cancellation of any insurance maintained by the Association.
- 6.7 Owner Caused Damages. If, due to the act or neglect of an Owner or such Owner's Agent, loss or damage shall be caused to any Person or property within the Common Areas, such Owner shall be liable and responsible for the payment of same. The amount of such loss or damage, together with costs of collection and reasonable attorneys' fees, if necessary, may be collected by the Board from such Owner as an Individual Assessment. Determination with respect to whether or not a particular activity or occurrence shall constitute a violation of this Section 6.7 shall be made by the Board and shall be final.
- 6.8 <u>Lease of a Lot</u>. An Owner shall have the right to lease his or her Lot upon such terms and conditions as the Owner may deem advisable, subject to the following:
- (a) Any such lease or rental agreement must be in compliance with applicable local, state and federal laws;
- (b) No Owner may lease or rent (i) less than his or her entire Lot, (ii) for transient or hotel purposes, or (iii) for a term of less than 3 months in duration except for month-to-month holdover terms after the initial term of the lease;
- (c) Any lease or rental agreement shall be in writing and shall provide that the lease or rental agreement is subject to the terms of the Governing Documents;
- (d) Such lease or rental agreement shall state that the failure of the lessee or renter to comply with the Governing Documents shall constitute a default and such default shall be

enforceable by either the Board or the lessor, or by both of them to include, but not be limited to, eviction of the lessee from the Lot; and

(e) The Board shall be furnished with a copy of the lease or rental agreement upon its request.

The Board can adopt additional Rules regarding the foregoing provisions of this Article Six.

ARTICLE SEVEN EASEMENTS

- 7.1 <u>Generally</u>. The Project shall be subject to all easements as shown or created on the Plat, those of record, those provided in the Act and those set forth in this Article Seven and in other provisions of this Declaration. The Board is hereby granted the right to establish from time to time utility and other easements, permits or licenses over the Common Areas for the best interest of the Association as a whole.
- 7.2 <u>Utility Easements</u>. There is hereby created and granted to Declarant, Owners, the Association and utility providers, a blanket easement on, over, in, under and through the Project for the installation, replacement, repair, operation and maintenance of utilities, including, but not limited to, water, sewer, gas, telephone, electricity and satellite and cable systems. Said blanket easement includes future utility services not presently available to the Project that may be reasonably required in the future.

Should any utility company furnishing a service covered by the easement herein created request a specific easement by separate recordable document, Declarant shall have, and hereby reserves, the right and authority to grant such easement upon, across, over or under any part or all of the Project without conflicting with the terms hereof; *provided*, *however*, that such power shall cease upon termination of the Declarant Rights, at which time such reserved right shall vest in the Association.

The easements granted in this Section 7.2 shall in no way affect, avoid, extinguish or modify any other recorded easement(s) within the Project.

7.3 Easements for Association and Owners. The Board (its agents, employees, and contractors) is hereby granted an easement on, over, in, under and through each Lot to perform its obligations pursuant to this Declaration, including, without limitation, as to the Association Maintenance Areas. Each Owner, and such Owner's Agents, is hereby granted a perpetual, non-exclusive right of ingress to and egress from the Owner's Lot, over and across the Common Areas, which right shall be appurtenant to the Owner's Lot, and which right shall be subject to limited and reasonable restrictions on the use of Common Areas set forth in writing by the Association, such as for closure for repairs and maintenance.

Each Owner, and such Owner's Agents, is hereby granted a perpetual, non-exclusive (a) right of pedestrian and vehicular ingress and egress over and across those portions of the private drives located within the Expansion Property that have been constructed but not yet annexed into the

Project, and (b) right to use the utilities and the storm detention facility located within the Expansion Property that have been constructed to serve such Owner's Lot, but not yet annexed into the Project. The foregoing easements shall automatically terminate at such time that the applicable private drive, utilities and/or storm detention facility is annexed into the Project.

- 7.4 <u>Emergency Easements</u>. A nonexclusive easement for ingress and egress is hereby granted to all police, sheriff, fire protection, ambulance, and other similar emergency agencies or persons, now or hereafter servicing the Project, to enter upon any part of the Project in the performance of their duties.
- 7.5 <u>Recording Data Regarding Easements</u>. Pursuant to §205(m) of the Act, the recording data for certain recorded easements and licenses appurtenant thereto, or included in the Project or to which any portion of the Project is or may become subject to as of the date of the recording of this Declaration, are identified on the attached <u>Exhibit C</u>.
- 7.6 Expansion Property Easement. Until such time (if ever) that all of the Expansion Property is annexed into the Project, the owner(s) of the Expansion Property shall have an easement over the Common Areas for (a) vehicular and pedestrian access, (b) construction purposes, and (c) the location, operation, maintenance repair and replacement of utilities serving the Expansion Property. This paragraph shall not be amended except with the written consent of the then-owner(s) of the Expansion Property.
- 7.7 <u>Easements Deemed Appurtenant</u>. The easements and rights herein created for an Owner shall be deemed appurtenant to the Lot owned by such Owner. All conveyances and instruments affecting title to a Lot shall be deemed to grant and reserve the easements and rights of way as provided herein, as though set forth in said document in full, even though no specific reference to such easements or rights of way appear.
- Tasements for Encroachments. If any part of a Home (or any improvement constructed upon the Expansion Property) as originally constructed by Declarant including, without limitation, patios, downspouts, fences and window wells encroaches or shall hereafter encroach upon the Common Areas, or upon another Lot, the Owner of that Home shall and does have an easement for the existence of such encroachment and for the maintenance of same. Such easements shall extend for whatever period of time the encroachment shall exist. Such encroachments shall not be considered to be encumbrances either upon the Common Areas or upon a Lot. Encroachments referred to herein include unintentional encroachments made by error in original construction of the Home (or any improvement constructed upon the Expansion Property), by the settling, rising or shifting of the earth, or by changes in position caused by repair or reconstruction of a Home (or any improvement constructed upon the Expansion Property).

ARTICLE EIGHT INSURANCE

8.1 <u>Association Insurance</u>.

- (a) Required Coverage. The Association, acting through the Board or its duly authorized agent, shall obtain and continue in effect the following types of insurance if reasonably available or, if not reasonably available, the most nearly equivalent coverages as are reasonably available. All polices of insurance purchased by the Association shall adhere to current Agency requirements, including, but not limited to, the carrier of such insurance meeting current Agency rating requirements, and notwithstanding the provisions of this Article Eight, the Board shall purchase such insurance in conformance with such Agency requirements.
- (i) The Association shall maintain blanket "all risk" property insurance in an amount not less than one hundred percent of the current replacement cost of the Common Areas. The Association's insurance may exclude land, excavations, foundations and other items normally excluded from such property insurance policies and may provide for a deductible in an amount not to exceed a reasonable and prudent amount as determined by the Board.
- (ii) Commercial general liability insurance, insuring the Association and the Owners against damage or injury caused by the negligence of the Association or any of its Owners, employees, agents or contractors while acting on its behalf. All Owners must be named as additional insureds for claims and liabilities arising in connection with the ownership, use or management of the Common Areas. If generally available at reasonable cost, the commercial general liability coverage (including primary and any umbrella coverage) shall have a limit of at least \$2,000,000.00 per occurrence with respect to bodily injury, personal injury and property damage.
- (iii) Workers' compensation insurance and employer's liability insurance to the extent required by law.
- (iv) Directors' and officers' liability coverage in an amount determined by the Board providing coverage for the members of the Board and having no exclusion for full coverage of Declarant appointed Board members.
- (v) Fidelity insurance covering all Persons responsible for handling Association funds in an amount determined in the Board's business judgment but not less than 3 months' Assessments plus all reserves on hand, and containing a waiver of all defenses based upon the exclusion of Persons serving without compensation.
 - (vi) Such additional insurance as the Board determines advisable.
- (b) Policy Requirements. All Association policies shall provide for a certificate of insurance to be furnished to the Association and upon request, to any Owner or Mortgagee. Premiums for all insurance maintained by the Association shall be a Common Expense and shall be included in the Common Expense Assessment. In the event of an insured loss, the deductible shall be treated as a Common Expense in the same manner as the premiums for the applicable insurance coverage. However, if the Board reasonably determines after providing Notice and Hearing, that the

loss is the result of the recklessness or willful misconduct of one or more Owners or an Owner's Agent, then the Association may specifically assess the full amount of such deductible against such Owner and their Lot as an Individual Assessment pursuant to Section 5.4(c) above. All insurance coverage obtained by the Association shall:

- (i) Be written with companies authorized to do business in the State of Colorado:
- (ii) Be written in the name of the Association as trustee for the Association and the Owners;
- (iii) Be written as a primary policy, not contributing with and not supplemental to the coverage that any Owner, Owner's Agent or Mortgagee may carry individually;
 - (iv) Include an inflation guard endorsement, as applicable;
- (v) Include an agreed amount endorsement, if the policy contains a co-insurance clause;
- (vi) Include an endorsement precluding cancellation, invalidation, suspension or non-renewal by the insurer on account of any curable defect or violation or any act or omission of any Owner, without prior written demand to the Association to cure the defect, violation, act or omission and allowance of a reasonable time to affect such cure;
- (vii) Include an endorsement precluding cancellation, invalidation or condition to recovery under the policy on account of any act or omission of any Owner, unless such Owner is acting within the scope of its authority on behalf of the Association; and
- (viii) Include an endorsement requiring at least 30-days prior written notice to the Association of any cancellation, substantial modification or non-renewal.
- (c) Other Policy Provisions. In addition, the Association may use reasonable efforts to secure insurance policies that provide:
- (i) A waiver of subrogation as to any claims against the Board, the officers or employees of the Association;
- (ii) A waiver of the insurer's rights to repair and reconstruct instead of paying cash;
- (iii) An endorsement excluding Owners' individual policies from consideration under any "other insurance" clause;
 - (iv) A cross liability provision; and
- (v) A provision vesting in the Association exclusive authority to adjust losses; provided, however, no Mortgagee having an interest in such losses may be prohibited from participating in settlement negotiations, if any, related to such losses.

- 8.2 <u>Separate Insurance</u>. No Owner shall be entitled to exercise his or her right to acquire or maintain such insurance coverage so as to decrease the amount which the Board, on behalf of all Owners, may realize under any insurance policy maintained by the Board or to cause any insurance coverage maintained by the Board to be brought into contribution with insurance coverage obtained by an Owner. All such policies shall contain waivers of subrogation.
- 8.3 Owner Insurance. Each Owner, at their sole cost and expense, shall (a) carry property insurance on all portions of their Lot and all Improvements thereon, and (b) carry general liability insurance providing coverage for bodily injury and property damage for the benefit of the Owner in amounts and with coverage as are from time to time customarily maintained by prudent owners of similar property.

ARTICLE NINE RESTORATION UPON DAMAGE OR DESTRUCTION/CONDEMNATION

- 9.1 <u>Duty to Restore Common Areas</u>. In the event of damage or destruction to any portion of the Common Areas which is covered by insurance carried by the Association, the insurance proceeds shall be applied by the Board to reconstruction and repair. The Common Areas must be repaired and restored in accordance with either the original plans and specifications, or other plans and specifications which have been approved by the Board.
- 9.2 <u>Use of Insurance Proceeds</u>. If the insurance proceeds with respect to such damage or destruction are insufficient to repair and reconstruct the damage to the Common Areas, the Board shall levy a Special Assessment in the aggregate amount of such insufficiency, and shall proceed to make such repairs or reconstruction. If all of the damage to the Common Areas covered by the Association's insurance is not repaired or reconstructed, the insurance proceeds attributable to the damage shall be used to restore the damaged portion of the Common Areas to a condition compatible with the remainder of the Project and the remainder of the proceeds shall be distributed to the Association.
- 9.3 <u>Condemnation</u>. If a part of the Common Areas is acquired by condemnation, that portion of any award attributable to the Common Areas taken must be paid to the Association as attorney-infact to be held in trust for the use and benefit of the Association, the Owners and their Mortgagees, as their interests may appear. No Owner or any other party shall be entitled to priority over Mortgagees with respect to any distribution of a condemnation award.

ARTICLE TEN MAINTENANCE

10.1 <u>Maintenance of Common Areas and Association Maintenance Areas</u>. The Association shall keep and maintain the Common Areas and Association Maintenance Areas in an attractive, clean and functional condition and in good repair and may make necessary or desirable alterations or improvements thereon. Determination of whether the foregoing repair or maintenance is the obligation of the Association, or if the repair and maintenance is necessary, shall rest solely with the Board, which will also have the sole responsibility for determining the kind and type of materials used in such repair and maintenance. **All Association maintenance, repair and replacement shall**

be undertaken in accordance with the maintenance manual(s) provided to the Association, if any.

- Maintenance of the Lots. Except for the Association Maintenance Areas described in the Rules and except as provided in Section 10.3 below, all Lot maintenance (including snow removal), repair and replacement shall be the obligation of the Owner of the Lot. Determination of whether maintenance of Lots is the obligation of the Association, or if the maintenance is necessary, shall rest solely with the Board, which will also have the sole responsibility for determining the kind and type of materials used in such maintenance. All Owner maintenance, repair and replacement shall be undertaken in accordance with the maintenance manuals provided to each original Owner of a Lot, if any.
- 10.3 Owner's Failure to Maintain and Repair. In the event that a Lot is not properly maintained and repaired by an Owner, and if the maintenance or repair responsibility for the unmaintained or unrepaired portion of the Lot lies with the Owner of the Lot, then the Board, after Notice and Hearing to the Owner (and after a determination by the Board that the condition of such Lot negatively impacts other Owners or the value of other Lots within the Project) shall have the right to enter upon the Lot to perform such work as is reasonably required to restore the Lot to a condition of good order and repair and charge the cost thereof to such Owner as an Individual Assessment.
- Maintenance of the Drainage Pattern/Irrigation System. Except for changes resulting from the initial construction of improvements undertaken by Declarant, there shall be no interference with the established drainage pattern initially established over any portion of the Project or the irrigation system installed within that Common Areas and/or a Lot (including systems and sprinklers), except as approved in writing by the Committee. The established drainage pattern may include the drainage pattern from the Common Areas over any Lots within the Project and from any Lot within the Project over the Common Areas or from any Lot over another Lot.
- 10.5 Owner-Caused Damages. If, due to the act or neglect of an Owner or an Owner's Agent, loss or damage shall be caused to any Person or property within the Common Areas or another Lot (or within the Owner's own Lot), such Owner shall be liable and responsible for the payment of same. The amount of such loss or damage, together with costs of collection and reasonable attorneys' fees, if necessary, may be collected by the Board, from such Owner as an Individual Assessment. Determination with respect to whether or not a particular activity or occurrence shall constitute a violation of this Section 10.5 shall be made by the Board and shall be final.

ARTICLE ELEVEN EXPANSION

11.1 Reservation of Right to Expand. Declarant reserves the right (without in any way being bound) to enlarge the Project without the necessity of the consent or joinder thereto, by the Owners, the Association, any Mortgagee or any other Person, by submitting to the Project from time to time all or any portion of the Expansion Property. In addition to the foregoing, Declarant reserves the right to annex real estate (in addition to the Expansion Property) to the Project from such locations as Declarant may elect in its sole discretion, so long as the total additional real estate so annexed to the

Project pursuant to this sentence, and not described on <u>Exhibit A</u> or <u>Exhibit B</u>, does not exceed 10% of the total area described in <u>Exhibit A</u> and <u>Exhibit B</u>.

- 11.2 <u>Supplemental Declarations</u>. Such expansion must be accomplished by recording a supplement(s) to the Declaration with the Clerk and Recorder containing the legal description of the new real property to be included in such expansion. The expansion may be accomplished in stages by successive supplements or in one supplemental expansion.
- 11.3 Expansion of Definitions. In the event of such expansion, the definitions used in this Declaration shall be expanded. For example, "Lot," "Common Areas," "Property" and "Project" shall mean the Lots, Common Areas, Property and the Project described herein plus any additional Lots and Common Areas added by any supplement, and reference to this Declaration shall mean this Declaration as supplemented. The new Lots and Common Areas shall be subject to all the terms, covenants, conditions and restrictions of this Declaration, as amended and supplemented.
- 11.4 <u>Interests on Enlargement</u>. An Owner at the time of their purchase of a Lot which has been annexed into the Project by a supplement to this Declaration shall be a Member of the Association. Such Owner shall be entitled to the same non-exclusive use of the Common Areas and the same voting privileges as the Owners of the initial property brought into the Project and shall be subject to the same Assessments, except as otherwise provided herein or in any supplement/amendment to this Declaration.

Whenever any additional Lots are brought into the Project, the Common Expense Liability of each Owner after such addition will change and shall be reallocated in accordance with Section 1.3(a) above.

11.5 <u>Termination of the Right of Expansion</u>. The rights to expand the Project pursuant to the provisions of this Article Eleven shall terminate at the option of Declarant as set forth in a recorded instrument executed by Declarant, but in any event shall terminate without further act or deed in accordance with the limitations set forth in Section 12.3 below.

ARTICLE TWELVE DECLARANT RIGHTS

- 12.1 <u>Reservation</u>. Notwithstanding any provision herein to the contrary, Declarant reserves for itself the following Declarant Rights which may be exercised without the necessity of the consent or joinder thereto, by the Owners, the Association, any Mortgagee or any other Person, where applicable, anywhere within the Project:
 - (a) To complete any improvements shown on the Plat;
- (b) To exercise any Declarant Rights reserved or described herein and as provided by the Act;
- (c) To maintain business/sales offices, sales trailers, parking spaces, management offices, storage areas, construction yards, signage, advertisements and model homes upon the Common Areas or any Lot owned by Declarant;

- (d) To have and use, and to permit others to have and use, easements through the Common Areas as may be reasonably necessary for construction within the Project and for the purpose of discharging Declarant's obligations under the Act and this Declaration;
- (e) To amend the Declaration and/or the Plat in connection with the exercise of any Declarant Rights, including, but not limited to, amending the Plat in order to create additional Lots and subdivide Lots owned by Declarant;
- (f) To expand, without in any way being bound, the Project in phases from time to time, by adding to the Project any of the Expansion Property and/or other unspecified property, which may consist of additional Common Areas, Homes and/or Lots, in accordance with Article Eleven above;
- (g) To merge or consolidate the Project with a common interest community of the same form of ownership;
- (h) To appoint or remove any officer of the Association or a member of the Board during the Period of Declarant Control, subject to the provisions of Section 4.8 above;
- (i) To designate certain Common Areas as Limited Common Areas by amendment to this Declaration;
- (j) To modify, amend or revise plans for construction, location and design of any Common Area (including any Improvements located thereon);
- (k) To grant and to have easements over the Common Areas to benefit any Lot and/or any property or portion of property that is contained within the Expansion Property whether or not such Expansion Property is annexed into the Project;
- (l) To withdraw from the Project any Lot which has not been conveyed to a consumer by Declarant;
 - (m) The rights described in Article Sixteen below;
- (n) To exercise any other "development rights" or "special declarant rights" as described in the Act; and
- (o) To exercise any other Declarant Right created or implied by any other provisions of this Declaration or the Act.
- Rights Transferable. Declarant Rights created or reserved under this Article Twelve for the benefit of Declarant may be transferred to any Person by Declarant by an instrument recorded with the Clerk and Recorder describing the Declarant Rights transferred. Such instrument shall be executed by the transferor Declarant and the transferee.
- 12.3 <u>Limitations</u>. Except as otherwise set forth in this Declaration (such as Article Sixteen below), Declarant Rights shall terminate at the option of Declarant as set forth in a recorded

instrument executed by Declarant, but in any event such Declarant Rights shall terminate without further act or deed 15 years after the date of the recording of this Declaration.

In the event that the process of entitlement for Declarant to obtain building permits is placed on "hold" (e.g., moratorium, anti-growth legislation, etc.) for reasons beyond the control of Declarant, the time limitations set forth herein shall be extended for a period of time equivalent to the period of time such impediment to entitlement was in effect.

- 12.4 <u>Interference with Declarant Rights</u>. Neither the Association, the Board nor any Owner may take any action or adopt any rule or regulation that will interfere with or diminish Declarant Rights without the prior written consent of Declarant.
- 12.5 <u>Declarant's Easements</u>. Declarant reserves for itself the right to (a) perform warranty work, and repair and construction work on Lots and Common Areas, (b) store materials in secure areas upon the Common Areas and/or Lots owned by Declarant, and (c) to control and have the right of access to work and repair until completion. All work performed by Declarant shall not require the consent or approval of the Board, Owners, Mortgagees or any other Person.

Declarant shall have an easement through the Common Areas as may be reasonably necessary for the purpose of discharging its obligations or exercising its rights, whether arising under the Act, reserved in this Article Twelve or reserved in other provisions of this Declaration. Notwithstanding any other provision of this Declaration (including, but not limited to, Section 12.3 above), the easements reserved herein shall remain in effect for the benefit of Declarant until the termination of all applicable warranty periods with respect to any particular Lot, Home or Common Areas.

12.6 Exercise of Rights. The exercise of any or all of the rights reserved to Declarant shall be at the sole option and discretion of Declarant. No assurances are made with respect to the boundaries of the Project or the parcels of real property that may be subject to Declarant Rights nor the order in which Declarant Rights may be exercised. Subject to Section 12.3 above, Declarant Rights may be exercised at any time and from time to time, and, if Declarant exercises any Declarant Rights, such rights may but need not, be exercised as to all or any other portion of the Project.

Notwithstanding anything in this Declaration to the contrary, no consent or agreement of, or notice to, the Association, any Owners, any Mortgagee or any other Person shall be required in order to allow Declarant to exercise any of its rights hereunder, provided such exercise otherwise complies with the applicable provisions of this Declaration and/or the Act.

ARTICLE THIRTEEN DESIGN REVIEW

13.1 Generally. Other than Improvements originally constructed by Declarant, each Improvement must be constructed, and may thereafter be removed, altered, repaired or modified, in accordance with this Article Thirteen. The strict application of the following limitations and restrictions in any specific case may be modified or waived in whole or in part by the Committee if such strict application would be unreasonable or unduly harsh under the circumstances. Any such modification or waiver must be in writing.

- 13.2 <u>Committee Approval of Improvements Required</u>. Other than Improvements originally constructed by Declarant, approval by the Committee shall be required prior to the commencement of the construction, alteration, repair, modification, expansion, addition, removal, demolition or destruction of any Improvements on any portion of the Project. The purchase of any Lot within the Project does not grant any implied guarantee of approval of any Improvement to be located thereon by the Committee.
- 13.3 <u>Membership of the Committee</u>. A "<u>Committee</u>" shall be formed to review and approve or disapprove plans for Improvements. The Committee shall consist of up to 3 members, the initial number and the members of which shall be determined by Declarant in its sole discretion. Declarant shall have the continuing right to appoint and reappoint the members of the Committee, which right shall terminate at the option of Declarant, but in any event shall terminate without further act or deed upon the later of (a) Declarant no longer owning any portion of the Property or the Expansion Property, or (b) Declarant no longer owning any portion of the Property and its right to annex property into the Project pursuant to Article Eleven above has expired, the provisions of Section 12.3 above notwithstanding. Thereafter, the Committee shall consist of 3 members, and the Board shall have the right to appoint the members of the Committee. Members of the Committee appointed by the Board must be Owners. Notwithstanding the foregoing, after the Declarant's right to appoint the Committee has expired, the Board may decide to act as the Committee.

Members of the Committee appointed by Declarant may be removed at any time by Declarant and shall serve until resignation or removal by Declarant. Members of the Committee appointed by the Board may be removed at any time by the Board, and shall serve for such term as may be designated by the Board or until resignation or removal by the Board.

- 13.4 <u>Address of the Committee</u>. The address of the Committee shall be that of the principal office of the Association.
- 13.5 <u>Submission of Plans/Design Review Fee</u>. Prior to the commencement of work to accomplish any proposed Improvement, the Person proposing to make such Improvement ("<u>Applicant</u>") shall submit to the Committee, at its offices, or at such other place as the Committee may designate, such descriptions, surveys, plot plans, drainage plans, elevation drawings, construction plans, specifications and samples of materials and colors as the Committee shall reasonably request, showing the nature, kind, shape, height, width, color, materials, and location of the proposed Improvement. The Applicant shall also submit all approvals required of the applicable governmental agencies for the Improvements to be undertaken.

The Committee may, in its guidelines or rules, provide for the payment of a reasonable fee to accompany each request for approval of any proposed Improvement. The Committee may provide that the amount of such fee shall be uniform for similar types of any proposed Improvements or that the fee shall be determined in any other manner, such as the estimated cost of the proposed Improvement. Said fee may be used to compensate any consultant as the Committee deems necessary to assist the Committee in the performance of its duties. Members of the Committee may be reimbursed for services rendered and for directly related out-of-pocket expenses.

Except as otherwise prohibited by law and except for Improvements constructed by Declarant, no Improvement of any kind shall be erected, altered, placed, or maintained within the Project unless and until the final plans, elevations, and specifications therefor have received written approval from the Committee as herein provided.

13.6 <u>Delegation/Waiver</u>. The Committee may, at its discretion, delegate to the Board any of the powers granted to it by this Article Thirteen by written notice to the Board, indicating what powers and authority are granted to the Board. Such delegation shall be effective from the date such notice is given.

The approval or consent of the Committee, any representative thereof, or the Board, to any application for architectural approval shall not be deemed to constitute a waiver of any right to withhold or deny approval or consent by the Committee, any representative thereof, or the Board, as to any application or other matters whatsoever as to which approval or consent may subsequently or additionally be required.

The Committee may waive or grant reasonable variances or adjustments to any provision of this Article Thirteen in the event there is a practical difficulty or unnecessary hardship.

- 13.7 <u>Criteria for Approval</u>. The question of reasonableness and good faith is the standard applicable in reviewing plans for approval by the Committee. The Committee's review and approval shall be limited solely to a determination as to whether or not the Improvements proposed are in conformance with applicable governmental regulations.
- 13.8 <u>Decision of the Committee</u>. The decision of the Committee shall be made within 30 days after receipt by the Committee of all materials required by the Committee unless such time period is extended by mutual agreement of the Committee and the Applicant. The Committee's decision shall be in writing and, if the decision is not to approve a proposed Improvement, the reasons therefor shall be stated. The decision of the Committee shall be promptly transmitted to the Applicant at the address furnished by the Applicant to the Committee.

A majority vote of the Committee shall constitute the action of the Committee.

The Committee shall report in writing to the Board all final actions of the Committee, if requested by the Board.

13.9 <u>Appeal to the Board</u>. If the Committee disapproves or imposes conditions on the approval of a proposed Improvement, the Applicant may appeal to the Board by giving written notice of such appeal to the Board and the Committee within 10 days after notice of such disapproval or conditional approval is given to the Applicant.

The Board shall hear the appeal with reasonable promptness after reasonable notice of such hearing to the Applicant and the Committee and shall decide, with reasonable promptness, whether or not the proposed Improvement or the conditions imposed by the Committee shall be approved, disapproved or modified.

If the Committee approves a proposed Improvement, any Owner impacted by the Committee's decision may appeal the approval to the Board by giving written notice of such appeal to the Board, the Committee and the Applicant within 10 days after such approval.

The Board shall hear the appeal with reasonable promptness after reasonable notice of such hearing to the Applicant, the Owner impacted and the Committee. The Board shall decide with reasonable promptness, whether or not the proposed Improvement's approval shall be upheld. The decision of the Board shall be final and binding on the parties concerned.

The appeal rights set forth in this Section 13.9 shall only be available after the period in which Declarant has the right to appoint the Committee has terminated.

- 13.10 <u>Failure of Committee to Act on Plans</u>. Any request for approval of a proposed Improvement shall be deemed approved, unless disapproval or a request for additional information or materials is transmitted to the Applicant in writing by the Committee within 30 days after the date of receipt by the Committee of all necessary materials, as determined by the Committee.
- 13.11 <u>Prosecution of Work after Approval</u>. After approval of any proposed Improvement, the proposed Improvement shall be accomplished as promptly and as diligently as possible and in complete conformity with the approval given by the Committee, including the description of the proposed Improvement, any materials submitted to the Committee in connection with the proposed Improvement and any conditions imposed by the Committee. Failure to complete any proposed Improvement within the timeframe set forth in the approval, if any, shall constitute a violation of this Article Thirteen unless extended by the Committee.
- 13.12 <u>Notice of Completion</u>. Upon completion of the Improvement, the Applicant shall deliver a written "Notice of Completion" to the Committee. Until the date of receipt of a Notice of Completion, the Committee shall not be deemed to have notice of completion of any Improvement.
- 13.13 <u>Inspection of Work</u>. The Committee, or its duly authorized representative, shall have the right to inspect any Improvement prior to, and/or after completion; *provided*, *however*, the right of inspection shall terminate 30 days after the Committee receives a Notice of Completion from the Applicant.
- 13.14 Letter of Noncompliance. If, as a result of inspections or otherwise, the Committee or Board finds that any Improvement has been done without obtaining the approval of the Committee or Board, or was not done in substantial compliance with the approval of the Committee, including the description and materials furnished to, and any conditions imposed by, the Committee or Board, or was not completed within the timeframe set forth in the approval, if any, the Committee or Board shall notify the Applicant in writing of the noncompliance ("Letter of Noncompliance"); which notice shall be given, in any event, within 30 days after the Committee or Board has inspected the Improvement, but in no event no later than 30 days after the Committee's or Board's receipt of such Applicant's Notice of Completion. The Letter of Noncompliance shall specify the particulars of the noncompliance and shall require the Applicant to take such action as may be necessary to remedy the noncompliance.

- 13.15 <u>Failure of Committee to Act after Completion</u>. If, for any reason other than the Applicant's act or neglect, the Committee fails to notify the Applicant of any noncompliance within 30 days after receipt by the Committee of a written Notice of Completion from the Applicant, the Improvement shall be deemed to be in compliance if the Improvement was, in fact, completed as of the date of the Notice of Completion.
- 13.16 Appeal to the Board of a Finding of Noncompliance. If the Committee gives any Letter of Noncompliance, the Applicant may appeal to the Board by giving written notice of such appeal to the Board and the Committee within 10 days after receipt by the Applicant of the Letter of Noncompliance.

If, after a Letter of Noncompliance, the Applicant fails to commence diligently to remedy such noncompliance, the Committee shall request a finding of noncompliance by the Board by giving written notice of such request to the Board and the Applicant within 30 days after delivery to the Applicant of a Letter of Noncompliance. In either event, the Board after Notice and Hearing shall decide, with reasonable promptness, whether or not there has been such noncompliance and, if so, the nature thereof.

The appeal rights set forth in this Section 13.16 shall only be available after the period in which Declarant has the right to appoint the Committee has terminated.

13.17 <u>Correction of Noncompliance</u>. If the Committee or the Board determines that a noncompliance exists, the Applicant shall remedy or remove the noncompliance within a period of not more than 30 days from the date of receipt by the Applicant of a Letter of Noncompliance or a decision of the Board (if a Letter of Noncompliance is appealed pursuant to Section 13.16 above). If the Applicant does not comply with the Letter of Noncompliance or the Board's ruling within such time period, the Committee or Board may, at its option, record a "Notice of Noncompliance" against the Lot on which the noncompliance exists, or may remove the non-complying Improvement or may otherwise remedy the noncompliance.

The Board may levy an Individual Assessment in accordance with Section 5.4(c) above against the Owner of such Lot for such costs and expenses incurred in enforcing this Section 13.17. The right of Declarant (during the period in which Declarant has the right to appoint the Committee) or the Board to remedy or remove any noncompliance shall be in addition to all other rights and remedies that Declarant or the Board may have at law, in equity or under the Governing Documents.

- 13.18 <u>Meetings of the Committee</u>. The Committee shall meet from time to time as necessary to perform its duties hereunder.
- 13.19 <u>No Implied Waiver or Estoppel</u>. No action or failure to act by the Committee or by the Board shall constitute a waiver or estoppel with respect to future action by the Committee or the Board.
- 13.20 <u>Estoppel Certificates</u>. The Board shall, upon the reasonable request of any interested Person and after confirming any necessary facts with the Committee, furnish a certificate with respect to the approval or disapproval of any Improvement or with respect to whether any Improvement was made in compliance herewith. Any Person, without actual notice to the contrary, shall be entitled to rely

on said certificate with respect to all matters set forth therein. Notwithstanding the foregoing, during the period that Declarant appoints the members of the Committee such certificate must come from Declarant and not the Board.

- 13.21 <u>No Liability for Committee Action</u>. There shall be no liability imposed on the Committee, any member of said Committee, any authorized representative of said Committee, the Association, any member of the Board or Declarant for any loss, damage or injury arising out of or in any way connected with the performance of the duties of the Committee, if such party acted in good faith and without malice.
- 13.22 <u>Exemptions for Declarant</u>. So long as Declarant Rights exist, Declarant shall be exempt from the provisions of this Article Thirteen.

THE PROVISIONS OF THIS ARTICLE THIRTEEN ARE IN ADDITION TO ANY DESIGN REVIEW REQUIREMENTS SET FORTH IN THE MASTER DECLARATION.

ARTICLE FOURTEEN MORTGAGEE PROVISIONS

- 14.1 <u>Notice to Eligible Mortgagees</u>. Eligible Mortgagees will be entitled to timely written notice of:
- (a) any material condemnation loss or any casualty loss which affects a material portion of the Common Areas or the Lot on which the Eligible Mortgagee holds a Mortgage;
- (b) any 60-day delinquency in the payment of Assessments or charges owed by the Owner of the Lot upon which the Eligible Mortgagee holds a Mortgage;
- (c) any lapse, cancellation, or material modification of any mandatory insurance policy or fidelity bond maintained by the Association; and
- (d) any termination of the common interest community in accordance with Section 17.4 below.
- 14.2 Action by Mortgagee. If this Declaration or any other Governing Document requires the approval of any Eligible Mortgagees then, the Association shall send a dated, written notice and a copy of any proposed amendment by certified or registered mail "return receipt" requested to such Eligible Mortgagee at its most recent address as shown on the recorded deed of trust or recorded assignment thereof, or as otherwise delivered by such Eligible Mortgagee to the Association. An Eligible Mortgagee that does not deliver to the Association a negative response within 60 days after the date it receives proper notice shall be deemed to have approved the proposed amendment.

ARTICLE FIFTEEN OWNER'S ACKNOWLEDGMENTS AND WAIVERS

15.1 <u>Molds.</u> Molds, mildew, fungi, bacteria and microbiologic organisms (collectively, "<u>Molds</u>") are present in soil, air and elsewhere in the environment. Molds can proliferate in various

environments, including, among others, damp areas such as bathrooms and within walls and partitions. Certain parties have expressed concerns about the possible adverse effects on human health from exposure to Molds. Due to various reasons, including the varying sensitivities of different individuals to various types of Molds and other contaminants, as of the date of this Declaration there currently exists no state or federal standards regarding acceptable levels of exposure to Molds. According to the Consumer Product Safety Commission and the American Lung Association, some diseases or illnesses have been linked with biological pollutants in the indoor environment, including some forms of Molds. However, it is believed that many of these conditions may also have causes unrelated to the indoor environment. Therefore, as of the date of this Declaration it is unknown how many potential health problems relate primarily or exclusively to indoor air quality or Molds. By acquiring a Lot within the Project, each Owner acknowledges and agrees that Declarant is not qualified and has not undertaken to evaluate all aspects of this very complex issue, and EACH OWNER ACKNOWLEDGES THAT DECLARANT MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, CONCERNING, THE PAST, CURRENT OR FUTURE, PRESENCE OR ABSENCE OF MOLDS IN OR IN THE VICINITY OF THEIR LOT OR IN ANY OTHER PORTION OF THE PROJECT. Declarant recommends that each Owner, at the Owner's expense, conduct their own investigation and consult with such experts as such Owner deems appropriate regarding the occurrence and effects of Molds, the potential sensitivity or special risk individuals who will occupy or use the Lot may have with respect to Molds, and methods to reduce or limit Molds within the Lot.

Noise Disturbances. The Lot and the Project are located in an urban location. Each Owner acknowledges that there may be a certain amount of unpredictable noise disturbances within the Project ("Noise Disturbances"). The Noise Disturbances may include, without limitation: (a) street noise from pedestrians and automobiles; (b) noise from adjacent Lots, roads and drives; (c) noise from outdoor or indoor art, food, music or other types of festivals in the vicinity of the Project; (d) noise from concerts, sporting events and other outdoor and indoor entertainment, performances and special events in the vicinity of the Project; and (e) noise from busy streets and highways.

Each Owner, by acquiring a Lot, acknowledges that the Noise Disturbances, and the impacts and disturbances generated by the Noise Disturbances, may occur in and around the Project, and may occur during daytime and nighttime and may be temporarily or permanently interrupted, discontinued or modified, in whole or in part, from time to time. Each Owner, by acquiring a Lot, forever waives and releases any actions or claims the Owner and its successors and assigns may have against Declarant which in any way arises out of the existence, occurrence, or the temporary or permanent interruption, discontinuance or modification of the Noise Disturbances and such impacts and disturbances.

15.3 No View Easement. Notwithstanding anything contained in this Declaration to the contrary, each Owner, by acquiring a Lot, acknowledges and agrees that there is no easement or other right, express or implied, for the benefit of any Owner or Lot for light, view or air included in or created by this Declaration or as a result of ownership of the Lot. Likewise, each Owner, by acquiring a Lot, acknowledges and agrees that any view, sight lines, or openings for light or air available from the Lot, or anywhere else on the Project, may be blocked or altered in whole or in part in the future by virtue of natural or unnatural causes, including, but not limited to, future construction or expansion of commercial or residential buildings or facilities. EACH OWNER, BY ACQUIRING A LOT.

HEREBY DISCLAIMS ANY AND ALL REPRESENTATIONS, WARRANTIES, OBLIGATIONS OR LIABILITIES CONCERNING EASEMENTS OR OTHER RIGHTS, WHETHER EXPRESS OR IMPLIED, FOR LIGHT, AIR, OR VIEW IN SUCH OWNER'S LOT AND/OR THE PROJECT. EACH OWNER, BY ACQUIRING A LOT, HEREBY ACCEPTS SUCH DISCLAIMER, AND AGREES THAT DECLARANT WILL NOT HAVE ANY OBLIGATION OR LIABILITY FOR, AND WAIVES ANY CLAIM AGAINST DECLARANT RELATED TO ANY LOSS OF LIGHT, AIR, OR VIEW THAT MAY AFFECT THE LOT OR THE PROJECT.

- 15.4 Security. NEITHER THE ASSOCIATION NOR DECLARANT WILL IN ANY WAY BE CONSIDERED AN INSURER OR GUARANTOR OF SECURITY WITHIN THE PROJECT, AND NEITHER THE ASSOCIATION NOR DECLARANT WILL BE HELD LIABLE FOR ANY LOSS OR DAMAGE BY REASON OF FAILURE TO PROVIDE ADEQUATE SECURITY, INEFFECTIVENESS OF SECURITY MEASURES UNDERTAKEN, OR ACTS OF THIRD PARTIES. ALL OWNERS AND OWNER'S AGENTS ACKNOWLEDGE THAT THE ASSOCIATION AND DECLARANT ARE NOT INSURERS OF SECURITY AT THE PROJECT AND THAT EACH OWNER AND OWNER'S AGENT ASSUMES ALL RISK OF LOSS OR DAMAGE TO PERSONS, TO LOTS, AND TO THE CONTENTS OF LOTS, AND FURTHER ACKNOWLEDGE THAT THE ASSOCIATION AND DECLARANT HAVE MADE NO REPRESENTATIONS OR WARRANTIES, NOR HAS ANY OWNER OR OWNER'S AGENT RELIED UPON ANY REPRESENTATIONS OR WARRANTIES EXPRESSED OR IMPLIED, INCLUDING ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR ANY PURPOSE, RELATIVE TO ANY SECURITY MEASURES RECOMMENDED OR UNDERTAKEN.
- Other Properties. Each Owner, by acquiring a Lot, acknowledges that other properties are located adjacent to and in the general vicinity of the Project ("Other Properties") and that the Other Properties may be constructed and developed pursuant to the land uses permitted by the County's and City's zoning ordinances, as well as any other governmental rules, regulations, or policies in effect now or in the future (collectively, "Ordinances"). Declarant does not make any representations concerning the planned uses of the Other Properties. Each Owner, by acquiring a Lot, further acknowledges that the zoning for the Project and the Other Properties is established and governed by the Ordinances. Any amendment to those Ordinances requires approval by the County and/or City. By acquiring a Lot, each Owner acknowledges that they have not relied upon any statements or representations regarding the Project or the Other Properties, including, without limitation, any representations made by Declarant (or any agent of Declarant), except for those statements and representations expressly set forth in this Declaration.

15.6 Soils Condition.

(a) Acknowledgment. THE SOILS WITHIN COLORADO CONSIST OF BOTH EXPANSIVE SOILS AND LOW-DENSITY SOILS WHICH MAY ADVERSELY AFFECT THE INTEGRITY OF A LOT OR A COMMON AREA IF SUCH LOT OR COMMON AREA IS NOT PROPERLY MAINTAINED. EXPANSIVE SOILS CONTAIN CLAY MINERALS WHICH HAVE THE CHARACTERISTIC OF CHANGING VOLUME WITH THE ADDITION OR SUBTRACTION OF MOISTURE, THEREBY RESULTING IN SWELLING AND/OR SHRINKING SOILS. THE ADDITION OF MOISTURE TO LOW-DENSITY SOILS CAUSES A

REALIGNMENT OF SOIL GRAINS, THEREBY RESULTING IN CONSOLIDATION AND/OR COLLAPSE OF THE SOILS.

- (b) Waiver of Liability of Declarant. BY TAKING TITLE TO A LOT, EACH OWNER ACKNOWLEDGES AND AGREES THAT SUCH OWNER HAS WAIVED AND WILL BE DEEMED TO HAVE WAIVED THE RIGHT TO ANY AWARD OF DAMAGES AGAINST DECLARANT FOR ANY LOSS OR DAMAGE TO ANY PORTION OF THE LOT OR THE COMMON AREAS CAUSED BY, RESULTING FROM OR IN ANY WAY CONNECTED WITH, SOIL CONDITIONS ON OR UNDER ANY COMMON AREAS, INCLUDING SPECIFICALLY THE PRESENCE OF EXPANSIVE SOILS.
- Inspection by Others; Waiver of Post Inspection Liability. It is hereby expressly understood and agreed by all Owners upon acquiring a Lot that Declarant relies upon governmental inspectors and other qualified subcontractors and tradesmen to inspect the construction of the Lots and the Common Areas in order to verify compliance with construction plans and with any and all building code requirements applicable to residential construction. Declarant and each Owner further expressly understands and agrees that, with respect to the Lots and the Common Areas, upon compliance with the inspections required by the local building department and the issuance of a certificate of occupancy by the responsible governmental agency, Declarant will be deemed to have used its best efforts to construct such Lots and Common Areas in substantial compliance with the construction plans and all applicable building code requirements. EXCEPT AS OTHERWISE MAY BE PROVIDED IN A PURCHASE AND SALE CONTRACT OR OTHER WRITTEN AGREEMENT BETWEEN DECLARANT AND AN OWNER, EACH OWNER, BY ACQUIRING A LOT, HEREBY KNOWINGLY AND WILLINGLY WAIVES AS AGAINST DECLARANT AND ITS OWNERS, AGENTS AND CONTRACTORS ANY AND ALL DEMANDS, CLAIMS, ACTIONS AND CAUSES OF ACTION AND ALL LIABILITY, LOSSES, DAMAGE, COSTS OR EXPENSES THAT HAVE BEEN OR MAY BE INCURRED IN ASSOCIATION THEREWITH, INCLUDING, WITHOUT LIMITATION, ATTORNEYS' FEES AND EXEMPLARY DAMAGES. WHICH ARISE FROM OR ARE RELATED TO ANY NONCOMPLIANCE OF THE LOTS OR THE COMMON AREAS WITH CONSTRUCTION PLANS OR BUILDING CODE REOUIREMENTS, WHICH NONCOMPLIANCE IS NEITHER SUBSTANTIAL NOR MATERIAL IN NATURE AND WHICH NONCOMPLIANCE IS DISCOVERED AFTER THE ISSUANCE OF A CERTIFICATE OF OCCUPANCY FOR, RESPECTIVELY, SUCH LOTS OR COMMON AREAS; AND ANY SUCH NONCOMPLIANCE WILL BE DEEMED UNINTENTIONAL WITH RESPECT TO DECLARANT AND ITS OWNERS, AGENTS AND CONTRACTORS. TO THE EXTENT THAT ANY NONCOMPLIANCE WITH CODES OR WITH THE CONSTRUCTION PLANS IS DISCOVERED WITH REGARD TO ANY LOT OR THE COMMON AREAS, THE PROVISIONS OF ARTICLE SIXTEEN WILL GOVERN SUCH MATTER.

IN THE EVENT THAT ANY PROVISIONS IN THIS ARTICLE FIFTEEN CONFLICT WITH ANY APPLICABLE FEDERAL OR COLORADO STATUTES WHICH PROVIDE NON-WAIVABLE LEGAL RIGHTS, INCLUDING, WITHOUT LIMITATION, THE COLORADO CONSTRUCTION DEFECT ACTION REFORM ACT OR THE COLORADO CONSUMER PROTECTION ACT, THEN THE NON-WAIVABLE TERMS OF SUCH STATUTE SHALL CONTROL.

ARTICLE SIXTEEN DISPUTE RESOLUTION

TO THE EXTENT PERMITTED BY LAW, THE PROVISIONS OF THIS ARTICLE SIXTEEN SHALL CONTROL OVER ANY CONTRARY ALTERNATIVE DISPUTE RESOLUTION PROCEDURES APPLICABLE TO THE PROJECT, INCLUDING THOSE SET FORTH IN THE MASTER DECLARATION.

- 16.1 <u>Definitions Applicable to this Article Sixteen</u>. For purposes of this Article Sixteen only, the following terms have the following meanings:
 - (a) "AAA" means the American Arbitration Association.
 - (b) "Claimant" means any Party having a Claim.
- (c) "Claim" means, except as excluded or exempted by the terms of this Article Sixteen (including Section 16.3 below), any claim, counterclaim, cross-claim, third-party claim, grievance or dispute between one Party and another, regardless of how it may have arisen or on what it might be based (including, but not limited to, damages, indemnity, subrogation or contribution), including, without limitation, disputes arising out of or related to, regardless of the theory of liability: (i) the interpretation, application or enforcement of any Governing Document or Limited Warranty; (ii) the location, size, planning, sale, marketing, development, design, construction, maintenance, repair and/or condition of the Lots, Homes, Common Areas and Project, including, without limitation, the soils of the Project; (iii) any statements, representations, promises, warranties, or other communications allegedly made by or on behalf of any Party relating to the foregoing; (iv) the Colorado Consumer Protection Act; and (v) damages or loss to, or the loss of, real or personal property or personal injury caused by a defect in the design or construction of the Lots, Homes, Common Areas and/or Project.
- (d) "Inspecting Party" means a Party causing an inspection of the Subject Property to be made.
- (e) "<u>Limited Warranty</u>" means a written limited warranty given to a Party related to a Lot.
- (f) "Party" means each of the following: (i) architects, engineers, contractors, subcontractors, developers, Declarant and affiliates of Declarant, builders, builder vendors, engineers and inspectors performing or furnishing the design, supervision, inspection, construction or observation of construction of any improvement to real property that is a part of the Project or any other party responsible for any part of the design, construction, repair or maintenance of any portion of the Project, and any of such parties' affiliates, and the officers, directors, partners, shareholders, members, managers, employees and servants of any of them (each a "Development Party" and collectively, the "Development Parties"); (ii) all Owners, Owner's Agents, the Association (including its directors and committee members) and all other Persons subject to this Declaration, their officers, owners, employees and agents; and (iii) any Person not otherwise subject to this Declaration who agrees to submit to this Article Sixteen.

- (g) "Respondent" means any Party against whom a Claimant asserts a Claim.
- (h) "<u>Subject Property</u>" means the property and all improvements thereon regarding which a Party contends that a Claim pertains, and/or property and all improvements thereon being inspected and/or repaired under the inspection and correction right in Section 16.4 below.
- (i) "<u>Termination of Mediation</u>" means a period of time expiring 20 days after a mediator has been agreed upon by the Parties or chosen by AAA if the Parties cannot agree, or within such other time as agreed to by the Claimant and Respondent in writing, and upon the expiration of which the Claimant and Respondent have not settled the Claim.

16.2 <u>Intent of Parties; Applicability of Article; and Applicability of Statutes of Limitations.</u>

- (a) Each Party agrees to work towards amicably resolving disputes, without the emotional and financial costs of litigation. Accordingly, each Party agrees to resolve all Claims only by using the procedures in this Article Sixteen (in the order set forth in Sections 16.4 and 16.5 below), and not litigation. Further, each Party agrees that the procedures in this Article Sixteen shall be the sole and exclusive remedial process that each Party shall have for any Claim. Should any Party commence litigation or any other action against any other Party in violation of this Article Sixteen, such action shall be dismissed and such Party shall reimburse all costs and expenses, including attorneys' fees and court costs, incurred by the other Party in such litigation or action within 10 days after written demand.
- (b) By accepting a deed for a Lot, each Owner agrees to be bound by and to comply with this Article Sixteen. The Association agrees to be bound by and to comply with this Article Sixteen.
- (c) The Parties agree that no Claim may be commenced after the date set forth in an applicable Limited Warranty, and if the Claim is not covered by such Limited Warranty, then when institution of legal or equitable proceedings based on such Claim would be barred by the applicable statute of limitation and/or statute of repose or as otherwise limited by this Article Sixteen.
- 16.3 <u>Exclusions from "Claim.</u>" Unless specifically exempted by this Article Sixteen, all Claims between any of the Parties shall be subject to the provisions of this Article Sixteen. Unless all Parties thereto otherwise agree in writing, "Claim" does not include the following and shall not be subject to the provisions of this Article Sixteen:
- (a) Any action by the Association to enforce the provisions of the Governing Documents (other than this Article Sixteen) against an Owner or Owner's Agent;
- (b) Any action by the Association to assess or collect any Assessments or to enforce or foreclose any Assessment Lien;
- (c) Any claim, grievance or dispute under a Limited Warranty that itself contains binding alternative dispute resolution procedures that preclude litigation and that the Parties agree to utilize and be bound by as a final resolution of the Claim;

- (d) Any action, suit or proceeding to compel arbitration of a Claim or to enforce any award or decision of an arbitration conducted in accordance with this Article Sixteen or to enforce the terms of any written settlement agreement of a Claim; and
 - (e) Any action pursuant to the provisions of this Declaration concerning mechanics liens.
- 16.4 <u>Notice; Right to Inspect and Correct; Mediation</u>. Before the earlier of, as applicable (a) the service of a Notice of Claim as described in Colorado's Construction Defect Action Reform Act ("<u>CDARA</u>"), or (ii) initiating arbitration under Section 16.5 below (each referred to herein as "<u>Commencing a Formal Claim</u>"), the Claimant shall first comply with the procedures set forth in this Section 16.4 in the order noted below.
- (i) *Notice*. **First**, the Claimant shall provide notice to everyone Claimant contends contributed to the alleged problem. The notice shall state plainly and concisely:

The nature of the Claim, including all Persons involved and each Respondent's role in the Claim;

The legal or contractual basis of the Claim (i.e., the specific authority out of which the Claim arises); and

The specific relief and/or proposed remedy sought.

- (ii) Right to Inspect and Correct. Second, if the Claim involves an alleged defect or damage to or duty to repair or replace any improvement or real property, then Claimant shall also provide Respondent, for a period of 60 days after delivery of the foregoing notice ("Inspection/Correction Period"), the right to access, inspect, correct the condition of, or redesign any portion of any improvement or real property allegedly containing a defect or damage or otherwise correct the alleged defect or damage. In exercising these inspection and correction rights, the Inspecting Party and Respondent shall:
- (1) Act carefully to avoid unreasonable intrusion on, or harm, damage or costs to the other Party including using its best efforts to avoid causing any damage to, or interference with, any improvements on the Subject Property at issue;
- (2) Attempt to minimize disruption or inconvenience to any Person who occupies the Subject Property;
- (3) Remove daily all debris caused by the inspection and remaining on the Subject Property; and
- (4) In a reasonable and timely manner, at the sole cost and expense of the Inspecting Party, promptly remove all equipment and materials from the Subject Property, repair and replace all damage, and restore the Subject Property to its pre-inspection condition unless the Subject Property is to be immediately repaired.

The Inspecting Party and Respondent shall not permit any lien arising from the inspection to attach to the Subject Property.

- (iii) Discussion of Claim. Third, in the event that (1) by the end of the Inspection/Correction Period described above, Respondent has elected not to access, inspect, correct the condition of, or redesign any portion of any improvement or real property allegedly containing a defect or damage or otherwise correct the alleged defect or damage, (2) by the end of the Inspection/Correction Period, Claimant is unsatisfied with actions undertaken by Respondent under Section 14.4(ii) above, or (3) the Claim does not involve an alleged defect or damage to any improvement or real property, then before Commencing a Formal Claim against any Respondent, the Parties shall first make every reasonable effort to meet in person and confer to resolve the Claim by good faith negotiation. The Parties shall seek to understand clearly the Claim and resolve as many aspects or issues as possible. Any Party may be represented by attorneys and independent consultants to assist such Party, including by attending all negotiations.
- (iv) *Mediation*. **Fourth**, if the Parties cannot resolve the Claim through negotiations under Section 16.4(iii) above after attempting to do so for 15 days, Claimant shall have an additional 10 days to submit the Claim to mediation under the auspices of AAA under the AAA's Commercial or Construction Industry Mediation Rules, as appropriate.
- (1) If Claimant does not submit the Claim to mediation within such time, or does not appear for the mediation, Claimant shall be deemed to have fully and finally waived the Claim for all purposes, such that Respondent shall be deemed released and discharged from all liability to Claimant for such Claim.
- (2) Any settlement of the Claim through mediation shall be documented in writing by the mediator and signed by the Parties. If mediation ends without a settlement, the mediator shall issue a notice of Termination of Mediation. This notice shall state that the Parties are at an impasse and the date that mediation was terminated.
- (3) Each Party shall pay its own costs of the mediation, including its own attorneys' fees. Each Party shall share equally all of the mediator's charges.
- (4) If the Parties resolve any Claim through negotiation or mediation under Section 16.4(iii) above or this Section 16.4(iv), and any Party later fails to comply with a written settlement agreement, then any other Party may file suit or initiate arbitration proceedings to enforce such written agreement without the need to again comply with such procedures. In such event, the Party taking action to enforce the agreement shall be entitled to recover from the non-complying Party (or if more than one non-complying Party, from all such Parties pro rata) all costs incurred in enforcing such agreement, including, without limitation, attorneys' fees and costs.

16.5 Commencing a Formal Claim.

(a) Only after receiving a notice of Termination of Mediation may a Claimant Commence a Formal Claim.

- (b) Before the Association may Commence a Formal Claim, the Board must first also comply with the following:
- (i) Provide written notice to all Owners and Respondent(s) calling a meeting to discuss the potential Claim ("<u>Homeowner Notice</u>"). The Homeowner Notice must be made in accordance with applicable State and local laws and also included in such notice shall be the following if not already required by such laws:
 - (1) The Approval Deadline (defined below);
- (2) If the Association were to prevail, what the Board expects that the Association may recover from the Respondent(s);
- (3) Whether the Board intends to enter into a contingency fee arrangement with the attorneys' representing the Association, and how much of the amount the Association recovers from the Respondent(s) will be paid to the attorney(s). What the Board estimates that, in addition to attorney fees, the Association will incur for consultants, expert witnesses, depositions, filing fees, and other expenses of pursuing the Claim;
- (4) If the Association makes a Claim and does not prevail, what the Board expects the Association will incur in witness and attorneys' fees and other costs;
- (5) If the Association does not recover from the Respondent(s), what it may have to pay to repair or replace any claimed defective work;
- (6) A statement that until any claimed defective work is repaired or replaced, or until the Claim is concluded, the market value of the affected Lots/Homes could be adversely affected;
- (7) A statement that until any claimed defective work is repaired or replaced, or until the Claim is concluded, Owners of affected Lots/Homes may have difficulty refinancing and prospective buyers of the affected Lots/Homes may have difficulty obtaining financing. In addition, a statement that certain federal underwriting standards or regulations prevent refinancing or obtaining a new loan in projects where a construction defect is claimed. In addition, a statement that certain lenders as a matter of policy will not refinance or provide a new loan in projects where a construction defect is claimed;
- (8) An estimate of the length of time it will take to reach a final resolution of the Claim (including appeals);
- (9) How the Association intends to finance the pursuit of the Claim (i.e., Special Assessments);
- (10) An affirmation from each Board member voting in favor of pursuing the Claim that the foregoing are true and correct; and

- (11) Any statement desired to be included in the notice by any Board member voting against pursuing the Claim.
- (ii) Require that repair estimates be given by contractors other than those recommended by the Association's attorneys.
- (iii) The Association meeting must be held no earlier than 10 days and no later than 15 days after the date of the Homeowner Notice. At the meeting the Respondent(s) must have an opportunity to address the Owners and the Board. Following the meeting and prior to the Association Commencing a Formal Claim, the Association must obtain the written approval to pursue the Claim from Owners of Lots to which a majority of the total votes in the Project (excluding votes allocated to Lots owned by Declarant if Declarant is a Respondent) are allocated. The Association must obtain such written Owner approval within 90 days after delivery of the Homeowner Notice or the Claim is deemed fully and finally released and may not be brought in any manner by the Association or the Owners ("Approval Deadline"). The votes of the Owners must be certified by an Association officer or agent, and written evidence of the certification shall be provided to the Respondent(s).
 - (c) Commencing a Formal Claim may only be accomplished by:
- (i) If the Claim is governed by CDARA, delivering a Notice of Claim under CDARA to Respondent(s). If the Parties fail to reach agreement on an offer of settlement pursuant to CDARA's Notice of Claim process (C.R.S. §13-20-803.5) and the Claimant elects to proceed with the Claim, then the Claim may proceed only by way of the arbitration procedures set forth below, and not by way of litigation.

Final, binding arbitration of the Claim shall be conducted under the auspices of AAA and its Commercial or Construction Industry Arbitration Rules, as appropriate. Claimant must provide to Respondent a "Notice of Intent to Arbitrate," within 20 days after the conclusion of the offer of settlement procedures set forth in C.R.S. §13-20-803.5. If Claimant does not timely initiate final, binding arbitration of the Claim and timely provide a Notice of Intent to Arbitrate to Respondent(s), then Claimant shall be deemed to have fully and finally waived the Claim, and Respondent(s) shall be deemed fully and finally released and discharged from all liability to Claimant for such Claim.

(ii) If the Claim is not governed by CDARA, then only by the arbitration procedures set forth below, and not by way of litigation. Final, binding arbitration of the Claim shall be conducted under the auspices of AAA and its Commercial or Construction Industry Arbitration Rules, as appropriate, in which event Claimant shall provide to Respondent a "Notice of Intent to Arbitrate," within 20 days after receiving the notice of Termination of Mediation. If Claimant does not timely initiate final, binding arbitration of the Claim and timely provide a Notice of Intent to Arbitrate to Respondent(s), then Claimant shall be deemed to have fully and finally waived the Claim, and Respondent(s) shall be deemed fully and finally released and discharged from all liability to Claimant for such Claim.

- (d) Mandatory Arbitration Procedures. The following arbitration procedures shall govern each arbitrated Claim:
- (i) The arbitrator must be a person qualified to consider and resolve the Claim with the appropriate industry and/or legal experience.
- (ii) No Person shall serve as the arbitrator where that Person has any financial or personal interest in the arbitration or any family, social or significant professional acquaintance with any Party to the arbitration. Any Person designated as an arbitrator shall immediately disclose in writing to all Parties any circumstance likely to affect the appearance of impartiality, including any bias or financial or personal interest in the arbitration ("<u>Arbitrator Disclosure</u>"). If any Party objects to the service of any arbitrator within 14 days after receipt of the Arbitrator's Disclosure, such arbitrator shall be replaced in the same manner as the initial arbitrator was selected.
- (iii) The arbitrator shall hold at least one hearing in which the Parties, their attorneys and expert consultants may participate. The arbitrator shall fix the date, time and place for the hearing. The arbitration proceedings shall be conducted at a mutually agreeable location in the City.
 - (iv) The arbitration shall be presided over by a single arbitrator.
- (v) Other than the deposition of experts and Claimant, no formal discovery shall be conducted without an order of the arbitrator or express written agreement of all Parties.
 - (vi) Unless directed by the arbitrator, there shall be no post-hearing briefs.
- (vii) The arbitration award shall address each specific Claim to be resolved in the arbitration, provide a summary of the reasons therefore and the relief granted, and be rendered no later than 14 days after the close of the hearing, unless otherwise agreed by the Parties. The arbitration award shall be in writing and shall be signed by the arbitrator.
- (viii) The arbitrator determines all issues about whether a Claim is covered by this Article Sixteen. Notwithstanding anything herein to the contrary (including, but not limited to, Section 16.5(d)(ix) and Section 16.5(d)(x) below), if a Party contests the validity or scope of arbitration in court, the arbitrator or the court shall award reasonable attorneys' fees and expenses incurred in defending such contests, including those incurred in trial or on appeal, to the noncontesting Party.
- (ix) The arbitrator shall apply the substantive law of Colorado with regard to any remedy granted. The arbitrator may award injunctive relief or any other remedy available in Colorado but shall not have the power to award punitive damages, attorneys' fees and/or costs to the prevailing Party. Except as set forth in Section 16.5(d)(viii) above, each Party is responsible for any fees and costs incurred by that Party. Any judgment upon the award rendered by the arbitrator may be entered in and enforced by any court of competent jurisdiction.
- (x) The Parties shall pay their pro rata share of all arbitration fees and costs, including, without limitation, the costs for the arbitrator and their consultants.

- (xi) The arbitrator shall have authority to establish reasonable terms regarding inspections, destructive testing and retention of independent consultants.
- (xii) Except as may be required by law or for confirmation of an arbitration award, and except as otherwise provided in this Article Sixteen, neither a Party nor an arbitrator may disclose the existence or contents of any arbitration without the prior written consent of all Parties to the arbitration.
- 16.6 <u>Notice of Certain Claims</u>. If a Claim includes a construction and/or design defect allegation, the Owner shall disclose the Claim and its details to his/her prospective purchasers and prospective Mortgagees.
- 16.7 Amendment. THE PROVISIONS OF THIS ARTICLE SIXTEEN INURE TO THE BENEFIT OF DECLARANT AND THE DEVELOPMENT PARTIES AND ALL OTHER PARTIES DESCRIBED IN THIS ARTICLE SIXTEEN, AND, NOTWITHSTANDING ANY OTHER PROVISION OF THIS DECLARATION, INCLUDING WITHOUT LIMITATION THE PROVISIONS OF ARTICLE SEVENTEEN BELOW, SHALL NOT BE AMENDED OR TERMINATED FOR A PERIOD OF 20 YEARS FROM THE DATE THIS DECLARATION IS RECORDED WITH THE CLERK AND RECORDER WITHOUT THE WRITTEN AND RECORDED CONSENT OF DECLARANT, WITHOUT REGARD TO WHETHER **DECLARANT OWNS** ANY LOT/HOME ATTHE TIME AMENDMENT/TERMINATION. BY TAKING TITLE TO A LOT/HOME, EACH OWNER ACKNOWLEDGES AND AGREES THAT THE TERMS OF THIS ARTICLE SIXTEEN ARE A SIGNIFICANT INDUCEMENT TO DECLARANT'S AND THE DEVELOPMENT PARTIES' WILLINGNESS TO DEVELOP AND SELL THE LOTS/HOMES AND THAT IN THE ABSENCE OF THE PROVISIONS CONTAINED IN THIS ARTICLE SIXTEEN, DECLARANT AND THE DEVELOPMENT PARTIES WOULD HAVE BEEN UNABLE AND UNWILLING TO DEVELOP AND SELL THE LOTS/HOMES FOR THE PRICES PAID BY THE ORIGINAL BUYERS.

IN ANY EVENT, ANY AMENDMENT TO OR DELETION OF ALL OR ANY PORTION OF THIS ARTICLE SIXTEEN SHALL NOT APPLY TO CLAIMS BASED ON ALLEGED ACTS OR OMISSIONS THAT PREDATE SUCH AMENDMENT OR DELETION.

THE TERMS OF THIS SECTION 16.7 SHALL NOT BE LIMITED BY THE PROVISIONS OF SECTION 12.3 ABOVE OR ANY OTHER PROVISION OF THIS DECLARATION.

16.8 Waiver of Jury Trial. IN THE EVENT THAT A COURT FINDS THAT THE DISPUTE RESOLUTION PROCEDURES SET FORTH IN THIS ARTICLE SIXTEEN ARE UNENFORCEABLE AND AS A RESULT A PARTY IS ALLOWED TO BRING A CLAIM IN COURT, ALL PARTIES AGREE THAT ANY LAWSUIT, WHETHER CLAIM OR COUNTERCLAIM, BROUGHT IN COURT SHALL BE TRIED ONLY BY A JUDGE AND NOT BY A JURY; AND EACH PARTY HEREBY KNOWINGLY, VOLUNTARILY, INTENTIONALLY, AND INTELLIGENTLY WAIVES, TO THE EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT TO A TRIAL BY JURY IN ANY SUCH SUIT.

16.9 <u>Conflict with Laws</u>. IN THE EVENT THAT THE PROVISIONS OF THIS ARTICLE SIXTEEN CONFLICT WITH ANY MANDATORY PROVISIONS OF APPLICABLE LAW, THE PROVISIONS OF THIS ARTICLE SIXTEEN SHALL BE REVISED TO THE MINIMUM EXTENT NECESSARY TO COMPLY WITH THE MANDATORY PROVISIONS OF SUCH LAWS.

ARTICLE SEVENTEEN DURATION, AMENDMENT, AND TERMINATION OF THE DECLARATION

- 17.1 <u>Duration</u>. The covenants, restrictions and obligations of this Declaration shall run with and bind the land in perpetuity until this Declaration is terminated in accordance with Section 17.4 below.
- 17.2 <u>Amendments by Owners</u>. Except for amendments that may be undertaken by the Declarant hereunder or pursuant to the Act, and except as otherwise provided for in Section 17.5 below or otherwise herein or pursuant to the Act, this Declaration may be amended by the written agreement of Owners of Lots to which at least 67% of the votes in the Association are allocated. Any amendment to Article Sixteen above shall also require the written consent of Declarant for a period of 20 years from the date this Declaration is recorded with the Clerk and Recorder. Any amendment of a material adverse nature to Mortgagees shall require the consent of Eligible Mortgagees representing 51% of the votes of Lots subject to such Mortgages.

Any such amendment shall be effective upon the recording of the amendment with the Clerk and Recorder together with a certificate certifying that the requisite number of Owners and Eligible Mortgagees (if applicable) have given their written consent (or deemed consent) to the amendment.

- 17.3 <u>Amendments by Declarant</u>. Declarant reserves the right to amend, without the consent of any Owner, the Association, any Mortgagee or any other Person, the Governing Documents at any time as follows:
- (a) To make nonmaterial changes, such as the correction of a technical, clerical, grammatical or typographical error or clarification of a statement;
 - (b) To comply with any requirements, standards or guidelines of any Agency; and
 - (c) To comply with any requirements of the Act or governmental agencies.
- 17.4 <u>Termination</u>. Notwithstanding Section 17.2 above, the Project may only be terminated upon the written agreement of: (a) Owners holding at least 67% of the vote in the Association; plus (b) Eligible Mortgagees representing 51% of the votes of Lots subject to such Mortgages, plus (c) Declarant for so long as it retains any Declarant Rights. Notwithstanding the foregoing, no termination of the Project shall affect the provisions of Article Sixteen above which shall survive any termination of the Project for a period of 20 years from the date this Declaration is recorded with the Clerk and Recorder.

17.5 <u>Consent Required</u>. So long as Declarant has any benefits, rights or obligations under or pursuant to this Declaration or any of the other Governing Documents, any proposed amendment/termination of any provision of the Governing Documents that in any manner affects such benefits, rights or obligations (including, but not limited to, Article Sixteen above) shall require Declarant's prior written consent to such amendment/termination. Obtaining Declarant's consent shall be required for the time period set forth in Section 12.3 above; *provided*, *however*, such time period limitations on Declarant's consent shall not apply to amendments/termination of Article Sixteen which instead shall be governed by the provisions of said Article. Any amendment made without such prior written consent as required herein shall be null and void and shall have no effect.

ARTICLE EIGHTEEN GENERAL PROVISIONS

Right of Action. Subject to the provisions of the Governing Documents (including, but not 18.1 limited to, Article Sixteen above), the Association on behalf of itself or 2 or more aggrieved Owners shall be granted a right of action against any and all Owners for failure to comply with the provisions of the Governing Documents, or with decisions of the Board made pursuant to authority granted to the Association in the Governing Documents. Subject to the provisions of the Governing Documents (including, but not limited to, Article Sixteen above), Owners shall have a right of action against the Association for failure to comply with the provisions of the Governing Documents, or with decisions of the Board made pursuant to authority granted to the Association in the Governing Documents. In any action covered by this Section 18.1, the Association or any Owner shall have the right, but not the obligation, to enforce the Governing Documents by any proceeding at law or in equity, or by mediation or binding arbitration (and the prevailing party in any such action shall be entitled to reimbursement from the non-prevailing party for all reasonable costs and expenses, including attorneys' fees, in connection with such action), except to the extent that (a) the notice and hearing procedures set forth in the Bylaws apply, or (b) the procedures set forth in Article Sixteen of this Declaration apply, or (c) any other agreement between Declarant and an Owner(s) applies. Failure by the Association or by any Owner to enforce compliance with any provision of the Governing Documents shall not be deemed a waiver of the right to enforce any provision thereafter.

Each Owner hereby irrevocably appoints the Board as the Owner's true and lawful attorney-in-fact for the purposes of submitting, executing and filing on behalf of Owners all documentation necessary to obtain all rezonings, site plans and/or replats that the Board feels necessary or prudent for the proper and effective operation of the Project. Acceptance by a grantee of a deed or other instrument of conveyance or any other instrument conveying a portion of the Property shall constitute appointment of the Board as the grantee's attorney-in-fact, and the Board shall have full authorization, right and power to make, execute and deliver any contract, assignment, deed, waiver or other instrument with respect to the interest of any Owner which may be necessary to exercise the powers granted to the Board as attorney-in-fact. In no event shall the Board's exercise of the foregoing rights affect Declarant Rights reserved hereunder and in the event that Declarant still retains Declarant Rights hereunder, the Board shall not exercise the foregoing rights without Declarant's prior written consent.

- 18.2 <u>Successors and Assigns</u>. This Declaration shall be binding upon and shall inure to the benefit of Declarant, the Association and each Owner and their heirs, personal representatives, successors and assigns.
- 18.3 <u>Severability</u>. The provisions of this Declaration shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Declaration, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) the invalid or unenforceable provision shall be reformed, to the minimum extent required to render such invalid or unenforceable provision enforceable in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision, and (b) the remainder of this Declaration and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision.
- 18.4 <u>No Waiver</u>. No provision contained in this Declaration shall be deemed to have been abrogated or waived by reason of any failure to enforce the same, irrespective of the number of violations or breaches which may occur.
- 18.5 Registration by Owner of Mailing Address; Notices. Each Owner shall register their mailing address with the Association. Except for monthly statements and other routine notices, which shall be personally delivered or sent by regular mail or as otherwise set forth in the Bylaws, all notices intended to be served upon an Owner pursuant to this Declaration, shall be delivered personally or sent by either registered or certified mail, postage prepaid, addressed in the name of the Owner at such registered mailing address or at the address of such Owner's Lot if there is no registered mailing address for such Owner on file at the Association.

All notices, demands or other notices intended to be served upon the Board or the Association shall be sent by certified mail, postage prepaid, to the Managing Agent, if any, and if there is no Managing Agent, then to the registered agent for the Association on file in the Office of the Secretary of State, State of Colorado.

- 18.6 <u>Conflicting Provisions</u>. The Governing Documents are intended to comply with the requirements of the Act and the Nonprofit Act (collectively, the "<u>Governing Acts</u>"). If there is any conflict between any provision of the Governing Documents and any mandatory provision of either of the Governing Acts, the mandatory provision of the applicable Governing Act shall control and neither Declarant, nor the Association shall have any liability for actions taken in conformity with such Governing Act. If there is any conflict between any provision of the Governing Documents and any permissive or non-mandatory provision of either of the Governing Acts, the provision of the Governing Documents shall control. In the event of any conflict between this Declaration and any other Governing Documents, this Declaration shall control, except as otherwise provided herein. In the event the Articles conflict with the Bylaws, the Articles shall control.
- 18.7 <u>Captions</u>. The captions and headings in this Declaration are for convenience only and shall not be considered in construing any provision of this Declaration.

18.8 <u>Numbers and Genders</u>. Whenever used herein, unless the context shall otherwise provide, the singular number shall include the plural, plural the singular, and the use of any gender shall include all genders.

first written above. Remington Homes at Candelas LLC, a Colorado limited liability company By: Printed Name: STATE OF) ss. **COUNTY OF** The foregoing instrument was acknowledged before me this 2019, by Guillaume Pouchot as of Remington Homes at Candelas LLC, a Colorado limited liability company limited liability company. CAROLYN PAYNE WITNESS my hand and official seal. Notary Public - State of Colorado Notary ID 20174026918 My Commission Expires Jun 27, 2021 My commission expires:

Notary Public

IN WITNESS WHEREOF, Declarant has caused this Declaration to be executed as of the date

MORTGAGEE CONSENT

Consent is hereby given to this Declaration by the undersigned mortgagee under the Deed of Trust recorded with the Clerk and Recorder on January 16, 2019 at Reception No. 2019004092. Additionally, the undersigned subordinates the lien and interests of the undersigned under said Deed of Trust as above referenced and under any other deeds of trust, financing statements or other security agreements for the benefit of the undersigned with regards to the Project to the covenants, terms and conditions of the Declaration, except that any such subordination relating to any lien created by this Declaration shall only be to the extent provided in the Act.

BOKF, NA dba Colorado State Bank and Trust	
By: 10-11-	
Printed Name: Darin Visscher	
Title: SVP	
STATE OF (closed)	
) ss.	
COUNTY OF (, / L M)	
Job La	
The foregoing instrument was acknowledged before me this & day of August 2019, t	ıν
Ownin Visschen as SVP of BOKF, NA d	-
Colorado State Bank and Trust.	U
Colorado State Dank and Trust.	
CAROLYN PAYNE	
WITNESS my hand and official seal. Notary Public – State of Colorado Notary Public – State of Colorado	
My commission expires: () () () My Commission Expires Jun 27, 2021	
My commission expires: (6) 27 2 My Commission Expires Jun 27, 2021	
(SEAL)	
Notary Public	

Exhibit A

(Property)

LOT 5B, LOT 6B, LOT 15B AND LOT 18B, BLOCK 21-B, CANDELAS FILING NO. 4, AMENDMENT NO. 1 FINAL PLAT, COUNTY OF JEFFERSON, STATE OF COLORADO

Exhibit B

(Expansion Property)

LOTS 1B THROUGH 22B, INCLUSIVE, BLOCK 13-B, CANDELAS FILING NO. 4, AMENDMENT NO. 1 FINAL PLAT, COUNTY OF JEFFERSON, STATE OF COLORADO

AND

LOTS 1B THROUGH 9B, INCLUSIVE, BLOCK 19-B, CANDELAS FILING NO. 4, AMENDMENT NO. 1 FINAL PLAT, COUNTY OF JEFFERSON, STATE OF COLORADO

AND

LOTS 1B THROUGH 13B, INCLUSIVE, BLOCK 20-B, CANDELAS FILING NO. 4, AMENDMENT NO. 1 FINAL PLAT, COUNTY OF JEFFERSON, STATE OF COLORADO

AND

LOT 3B, LOT 4B, LOTS 7B THROUGH 14B, INCLUSIVE, LOT 16B, LOT 17B AND LOTS 19B THROUGH 38B, INCLUSIVE, BLOCK 21-B, CANDELAS FILING NO. 4, AMENDMENT NO. 1 FINAL PLAT, COUNTY OF JEFFERSON, STATE OF COLORADO

Exhibit C

(Recording Data for Certain Recorded Easements, Licenses and Other Matters of Record which the Project is or may become Subject to)

- 1. RIGHT OF THE PROPRIETOR OF A VEIN OR LODE TO EXTRACT AND REMOVE HIS ORE THEREFROM, SHOULD THE SAME BE FOUND TO PENETRATE OR INTERSECT THE PREMISES HEREBY GRANTED, AND A RIGHT OF WAY FOR DITCHES OR CANALS CONSTRUCTED BY THE AUTHORITY OF THE UNITED STATES, AS RESERVED IN THE FOLLOWING UNITED STATES PATENTS: RECORDED JANUARY 19, 1881 IN BOOK 22 AT PAGE 294 (N 1/2 NE 1/4 AND N 1/2 NW 1/4 SECTION 22).
- 2. TERMS, CONDITIONS, PROVISIONS, BURDENS, OBLIGATIONS AND EASEMENTS AS TO ANY AND ALL OIL, GAS, COAL, AND OTHER MINERALS, AS CONVEYED, RESERVED AND/OR SET FORTH IN INSTRUMENTS RECORDED JULY 26, 1950 IN BOOK 682 AT PAGE 561, FEBRUARY 4, 1983 UNDER RECEPTION NO. 83010841, OCTOBER 3, 1985 UNDER RECEPTION NO. 85095067, OCTOBER 29, 1985 UNDER RECEPTION NO. 85104111, DECEMBER 6, 1990 UNDER RECEPTION NO. 90103378, APRIL 6, 1993 UNDER RECEPTION NO. 93045057, APRIL 17, 1998 UNDER RECEPTION NO. F0594290, NOVEMBER 5, 1998 UNDER RECEPTION NO. F0729050, JANUARY 27, 2005 UNDER RECEPTION NO. F2164505, FEBRUARY 15, 2006 UNDER RECEPTION NO. 2006019024, MAY 11, 2006 UNDER RECEPTION NO. 2006056338, MAY 11, 2006 UNDER RECEPTION NO. 2006056339, JUNE 22, 2006 UNDER RECEPTION NO. 2006075745, AUGUST 15, 2006 UNDER RECEPTION NO. 2006099768, JUNE 12, 2008 UNDER RECEPTION NO. 2008057659 AND SEPTEMBER 30, 2010 UNDER RECEPTION NO. 2010086272.

RELINQUISHMENT OF SURFACE RIGHTS IN CONNECTION THEREWITH

- 3. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE NORTH JEFFCO METROPOLITAN RECREATION DISTRICT, AS EVIDENCED BY INSTRUMENT RECORDED NOVEMBER 13, 1956, IN BOOK 1029 AT PAGE 504.
- 4. TERMS, CONDITIONS, PROVISIONS, BURDENS, OBLIGATIONS AND EASEMENTS AS TO ANY AND ALL WATER RIGHTS, AS CONVEYED, RESERVED AND/OR SET FORTH IN INSTRUMENTS RECORDED MAY 21, 1987 UNDER RECEPTION NOS. 87067102 AND 87067126, JANUARY 22, 1996 UNDER RECEPTION NO. F0174068, SEPTEMBER 2, 1997 UNDER RECEPTION NO. F0469594, JUNE 5, 2003 UNDER RECEPTION NOS. F1769738, F1769739 AND F1769740, MAY 11, 2006 UNDER RECEPTION NO. 2006056337 AND MAY 24, 2006 UNDER RECEPTION NO. 2006062063.
- 5. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE COAL CREEK CANYON FIRE PROTECTION DISTRICT, AS EVIDENCED BY INSTRUMENT RECORDED JUNE 01, 1987, UNDER RECEPTION NO.

87071382.

- 6. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN ANNEXATION ORDINANCE #89-36 RECORDED JANUARY 17, 1991 UNDER RECEPTION NO. 91004720 AND THAT ANNEXATION MAP RECORDED JANUARY 17, 1991 UNDER RECEPTION NO. 91004721 AND THAT ORDINANCE AMENDING THE LEGAL DESCRIPTIONS RECORDED APRIL 18, 1991 UNDER RECEPTION NO. 91032466.
- 7. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN ANNEXATION AGREEMENT RECORDED JANUARY 17, 1991 UNDER RECEPTION NO. 91004730.
- 8. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN SUBDIVIDER'S AGREEMENT RECORDED JANUARY 17, 1991 UNDER RECEPTION NO. 91004731.
- 9. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN INTERGOVERNMENTAL AGREEMENT RECORDED JANUARY 17, 1991 UNDER RECEPTION NO. 91004732.
- 10. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE NORTH JEFFCO PARK AND RECREATION DISTRICT, AS EVIDENCED BY INSTRUMENT RECORDED NOVEMBER 20, 2000 UNDER RECEPTION NO. F1146422.
- 11. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN COUNCIL BILL #99-036, ORDINANCE #3543 RECORDED JULY 22, 1999 UNDER RECEPTION NO. F0913069.
- 12. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PSM PROPERTIES, INC. LAND SURVEY PLAT RECORDED OCTOBER 27, 1999 UNDER RECEPTION NO. F0967966.
- 13. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE ARVADA FIRE PROTECTION DISTRICT, AS EVIDENCED BY INSTRUMENTS RECORDED JANUARY 25, 2001, UNDER RECEPTION NO. F1174251, MAY 16, 2001 UNDER RECEPTION NO. F1237241 AND JULY 12, 2011 UNDER RECEPTION NO. 2011064250.
- 14. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN REZONING ORDINANCE #3960 RECORDED AUGUST 04, 2005 UNDER RECEPTION NO. 2005060319.

- 15. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN DEVELOPMENT AGREEMENT PERTAINING TO VESTED RIGHTS FOR VAUXMONT RECORDED NOVEMBER 16, 2005 UNDER RECEPTION NO. 2005115232 AND FIRST AMENDMENT RECORDED SEPTEMBER 20, 2010 UNDER RECEPTION NO. 2010081848, AND SECOND AMENDMENT RECORDED JULY 30, 2015 UNDER RECEPTION NO. 2015079968.
- 16. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF VAUXMONT MINOR SUBDIVISION RECORDED JANUARY 04, 2006 UNDER RECEPTION NO. 2006000891.
- 17. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN INDEMNIFICATION AGREEMENT RECORDED MAY 11, 2006 UNDER RECEPTION NO. 2006056332.
- 18. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF VAUXMONT MINOR SUBDIVISION NO. 2 RECORDED APRIL 16, 2007 UNDER RECEPTION NO. 2007042669.

AFFIDAVIT OF STREET NAME CHANGE RECORDED JULY 3, 2008 UNDER RECEPTION NO. 2008065186.

RATIFICATION AND CONFIRMATION OF PLAT RECORDED SEPTEMBER 16, 2013 UNDER RECEPTION NO. 2013113342.

19. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE VAUXMONT METROPOLITAN DISTRICT, AS EVIDENCED BY INSTRUMENT RECORDED MARCH 31, 2009, UNDER RECEPTION NO. 2009027635.

JOINT RESOLUTION OF CIMARRON METROPOLITAN DISTRICT AND VAUXMONT METROPOLITAN DISTRICT REGARDING THE IMPOSITION OF A TRANSFER AND PLAN REVIEW FEE RECORDED APRIL 23, 2015 UNDER RECEPTION NO. 2015038951, AND FIRST AMENDMENT THERETO RECORDED JULY 22, 2015 UNDER RECEPTION NO. 2015076444, AND AMENDED AND RESTATED JOINT RESOLUTION RECORDED FEBRUARY 4, 2016 UNDER RECEPTION NO. 2016011068.

AMENDED AND RESTATED VAUXMONT METROPOLITAN DISTRICT RESOLUTION FOR THE IMPOSITION OF FEES FOR THE CENTRALIZED PROVISION OF TRASH REMOVAL AND RECYCLING SERVICES RECORDED DECEMBER 23, 2015 UNDER RECEPTION NO. 2015135696.

20. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN AMENDED AND RESTATED AGREEMENT FOR LEASE AND DONATION AND DEDICATION OF LAND RECORDED JANUARY 06, 2011 UNDER RECEPTION NO.

2011001924.

- 21. FEES, TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN JOINT RESOLUTION OF CIMARRON METROPOLITAN DISTRICT AND VAUXMONT METROPOLITAN DISTRICT REGARDING THE IMPOSITION OF CAPITAL FEES RECORDED APRIL 01, 2011 UNDER RECEPTION NO. 2011032905 AND AMENDED AND RESTATED JOINT RESOLUTION RECORDED AUGUST 6, 2012 UNDER RECEPTION NO. 2012082334, AND FIRST AMENDMENT TO AMENDED AND RESTATED JOINT RESOLUTION RECORDED APRIL 8, 2014 UNDER RECEPTION NO. 2014026279, AND SECOND AMENDMENT RECORDED NOVEMBER 2, 2015 UNDER RECEPTION NO. 2015117097 AND RESOLUTION NO. 2017-10-01 FOURTH AMENDMENT TO AMENDED AND RESTATED JOINT RESOLUTION OF CIMARRON METROPOLITAN DISTRICT AND VAUXMONT METROPOLITAN DISTRICT REGARDING THE IMPOSITION OF CAPITAL FEES RECORDED FEBRUARY 28, 2018 UNDER RECEPTION NO. 2018018482.
- 22. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF CANDELAS FILING NO. 1 RECORDED APRIL 21, 2011 UNDER RECEPTION NO. 2011039877.

AFFIDAVIT OF STREET NAME CHANGE RECORDED APRIL 19, 2012 UNDER RECEPTION NO. 2012041236 AND CORRECTED STREET NAME SPELLING RECORDED MAY 10, 2012 UNDER RECEPTION NO. 2012049318.

- 23. RESTRICTIVE COVENANTS, WHICH DO NOT CONTAIN A FORFEITURE OR REVERTER CLAUSE, BUT OMITTING ANY COVENANTS OR RESTRICTIONS, IF ANY, BASED UPON RACE, COLOR, RELIGION, SEX, SEXUAL ORIENTATION, FAMILIAL STATUS, MARITAL STATUS, DISABILITY, HANDICAP, NATIONAL ORIGIN, ANCESTRY, OR SOURCE OF INCOME, AS SET FORTH IN APPLICABLE STATE OR FEDERAL LAWS, EXCEPT TO THE EXTENT THAT SAID COVENANT OR RESTRICTION IS PERMITTED BY APPLICABLE LAW, AS CONTAINED IN INSTRUMENT RECORDED MARCH 26, 2012, UNDER RECEPTION NO. 2012032029 AND FIRST AMENDMENT RECORDED SEPTEMBER 10, 2012 UNDER RECEPTION NO. 2012095878, AND SECOND AMENDMENT RECORDED NOVEMBER 21, 2014 UNDER RECEPTION NO. 2014100102, AND THIRD AMENDMENT RECORDED SEPTEMBER 9, 2015 UNDER RECEPTION NO. 2015096329.
- 24. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE CANDELAS SPECIAL IMPROVEMENT DISTRICT NO. 1, AS EVIDENCED BY INSTRUMENT RECORDED MARCH 26, 2012, UNDER RECEPTION NO. 2012032030.
- 25. ANY TAX, LIEN, FEE, OR ASSESSMENT BY REASON OF INCLUSION OF SUBJECT PROPERTY IN THE APEX PARK AND RECREATION DISTRICT, AS EVIDENCED BY INSTRUMENT RECORDED JUNE 27, 2013, UNDER RECEPTION NO. 2013078081.

26. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF CANDELAS FILING NO. 3 RECORDED SEPTEMBER 26, 2013 UNDER RECEPTION NO. 2013117890.

AFFIDAVIT OF PLAT CORRECTION RECORDED DECEMBER 18, 2013 UNDER RECEPTION NO. 2013144893.

27. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF CANDELAS FILING NO. 4 RECORDED AUGUST 28, 2014 UNDER RECEPTION NO. 2014072177.

SURVEYOR'S AFFIDAVIT OF CORRECTION IN CONNECTION THEREWITH RECORDED DECEMBER 1, 2017 UNDER RECEPTION NO. 2017123489.

- 28. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN CANDELAS FILING FOUR SUBDIVIDER'S AND DEVELOPMENT AGREEMENT RECORDED AUGUST 28, 2014 UNDER RECEPTION NO. 2014072178, AND AS AMENDED IN INSTRUMENT RECORDED OCTOBER 12, 2017 UNDER RECEPTION NO. 2017105259.
- 29. TERMS, CONDITIONS, PROVISIONS, BURDENS AND OBLIGATIONS AS SET FORTH IN RESOLUTION NO: 2016-11-02 RECORDED NOVEMBER 29, 2016 UNDER RECEPTION NO. 2016125096.
- 30. EASEMENTS, CONDITIONS, COVENANTS, RESTRICTIONS, RESERVATIONS AND NOTES ON THE PLAT OF CANDELAS FILING NO. 4, AMENDMENT NO. 1 RECORDED OCTOBER 12, 2017 UNDER RECEPTION NO. 2017105258.